

# Puerto Rico and Section 936: A Costly Dependence

by J. Tomas Hexner and Glenn P. Jenkins

Reprinted from 10 Tax Notes Int'l 235 (January 16, 1995)

#### tax notes international

Copyright, 1995, Tax Analysts ISSN 1058-3971

#### Editor MaryGael Timberlake

#### International Tax Editor

John Turro

**Managing Editor** Julianne MacKinnon

#### Tax Notes International Staff

Associate Editor: Robert J. Wells

Assistant Editors: Sarah Barth and John Iekel Editorial Assistant: Violeta Kovaçevic

Production: Paul Doster Publisher: Thomas F. Field

#### Advisory Board

Chairman: Oliver Oldman (U.S.) Members: Brian J. Arnold and Richard M. Bird (Canada), Richard Goode (U.S.), Hiroshi Kaneko (Japan), Rick Krever (Australia), Steven R. Lainoff (U.S.), Michael J. McIntyre (U.S.), Leif Mutén (Sweden), H. David Rosenbloom (U.S.), and Frans Vanistendael (Belgium)

#### Senior Correspondents

Argentina: Guillermo Teijeiro, Negri, Teijeiro & Incera, Buenos Aires

Austratia: Lee Burns, Univ. of Sydney Faculty of Law, Sydney; Richard Vann, Univ. of Syndey Faculty of Law, Sydney

Belgium: Patrick Kelley, De Bandt, Van Hecke & Lagae, Brussels

Canada: Brian J. Arnold, Univ. of Western Ontario Faculty of Law/Goodman Phillips & Vineberg,

London, Ontario

Finland: Timo Viherkenttä, Ministry of Finance, Helsinki France: Olivier Delattre, Stibbe Simon Monahan Duhot, Paris

any: Dr. Jörg-Dietrich Kramer, Federal Academy of Finance, Bonn

Hong Kong: Jefferson P. VanderWolk, Baker & McKenzie, Hong Kong India: Nishith M. Desai, Bombay

Ireland: Mary Walsh, Coopers & Lybrand, Dublin Italy: Guelielmo Maisto, Maisto e Miscali, Milano

Itany: Gugiteimo Maisto, Maisto e Miscali, Milano Japan: Gary Thomas, White & Case, Tokyo Netherlands: Maarten J. Ellis, Loyens & Volkmaars, Rotterdam; Prof. Kees van Raad, The Hague Peru: Alex Morris, Rodrigo, Elias & Medrano, Abogados, Lima Poland: Dr. Janusz Fiszer, Warsaw University/White & Case, Warsaw Sweden: Prof. Leif Mutén, Stockholm School of Economics, Stockholm

Swetzerfand: Walter H. Boss, Pestalozzi Gmuer & Patry, Zurich
U.K.: John Goldsworth, Wendens Ambo, Essex, Jill C. Pagan, Barrister, London
U.S.: Richard Doemberg, Emory Univ. School of Law, Adanta GA., James Fuller, Fenwick & West, Palo
Alto; Philip Morrison, Baker & McKenzie, Washington, D.C.; Richard D. Pomp, Univ. Connecticut
School of Law, Hartford, CT.

#### Correspondents

Albania: Robert Ceku, Ministry of Finance, Tirana; Adriana Civici, Ministry of Finance, Tirana Angola: Trevor Wood, Ernst & Young, Lisbon

Anguilla: Laura Dempsey, Ministry of Finance, Anguilla Antigua: Donald B. Ward, Price Waterbouse Center, St. John's

Argentina: Ruben Asorey, Asorey & Navarrine, Buenos Aires; Nicasio del Castillo and Alberto R. Lopez, Coopers & Lybrand, New York; Horacio Crespo, Coopers & Lybrand, Buenos Aires

Armenia: Gagik Yeghiazarian, Arthur Andersen, Mosco

Australia: Arumuganathan Sabapathy, Adelaide, Timothy W. Cox, Price Waterhouse, Melbourne Austria: Friedrich Rödler, Price Waterhouse AG, Vienna

Austria: Friedrich Router, Frei Waternouse AA, Vietna Bahamas: Hywel Jones, Canadian Imperial Bank of Commerce Trust Company (Bahams) Ltd., Nassau Barbados: Patrick B. Toppin, Pannell Kerr Forster, Christ Church Belgium: Pascal Faes, Van Bael & Bellis, Brussels; Jan De Decker, Ernst & Young, Brussels Bermuda: Wendell Hollis, Mello, Hollis, Jones & Martin, Hamilton Bolivia: Gonzalo Ruiz Ballivián, KPMG Peat Marwick, La Paz

Botswana: I.O. Sennanyana, Deputy Director, Tax Policy, Ministry of Finance & Development Planning,

Brazil: Teresa Cristina Nogueira de Mello, Almeida, Mello e Braga, São Paulo
British Virgin Islands: William L. Blum, Counsel to Grunert Stout Bruch & Moore, St. Thomas, U.S.V.I.

Bulgaria: Todor Tabakov, Sofia

Cameroon: Edwin N. Forlemu, International Tax Program, Harvard University, Cambridge

Cayman Islands: Timothy Ridley, Maples & Calder, Grand Cayman

Chile: León Larrain, Cruzat, Ortuzar & MacKenna, Santiago Colombia: Leif Weizman, Bogotá

Cook Islands: David R. McNair, Southpac Trust Limited, Rarotonga Costa Rica: Humberto Pacheco, Pacheco Coto, San José

Cyprus: Andreas Gregoriou, Director, Inland Revenue Department of Cyprus, Nicosia; Theodoros Philippou, Price Waterhouse, Nicosia

Czech Republic: David Roach, Price Waterhouse, Prague; Oliver Brahmst, Squire Sanders & Dempsey,

Denmark: Aage Michelsen, Department of Business Law, The Aarhus School of Business, Aarhus V; Leif Weizman, Bogotá

Dominican Republic: Dr. Fernando Ravelo Alvarez, Santo Domingo Ecuador: Miguel Andrade, Estudio Juridico Teran & Teran, Quito

Egypt: Farouk Metwally, Senior Tax Consultant, Ernst & Young, Cairo Estonia: Peep Kalamäe, Head of International Taxation Division, Ministry of Finance, Tallinn

European Union: Pascal Faes, Van Bael & Bellis, Brussels; Nigel Tutt, Brussels, Belgium Fiji: Brian Murphy, KPMG Peat Marwick, Suva

France: Clotilde Fournier, Bureau Francis Lefebvre, New York

Republic of Gabon: Stéphane Brabant, Price Waterhouse, Libreville The Gambia: Samba Ebrima Saye, Income Tax Office, Banjul

Germany: Richard G. Minor, Baker & McKenzie, Washington DC; Dr. Karin L. Pilny, Feddersen Laule Scherzberg & Ohle Hansen Ewerwahn, Frankfurt am Main

Ghana: Seth Terkper, Ministry of Finance, Accra Gibraltar: Jonathan S. Baker, Ernst & Young, Gibraltar

Greece: Alexandra Gavrielides, Athens

Guatemala: Dr. Rodrigo Toriello, Bonilla Montano y Toriello, Guatemala City

Guernsey: Neil Crocker, Coopers & Lybrand, St. Peter Port Guyana: Lancelot A. Atherly, Georgetown

Hong Kong: Michael Olesnicky, Baker & McKenzie, Hong Kong Hungary: John Balazs, and Peter Gerendasi, Price Waterhouse, Budapest

Iceland: Indridi H. Thorlaksson, Reykjavik

India: Gagan Kumar Kawatra, Federation of Indian Chambers of Commerce and Industry, New Delhi Indonesia: Hadiyanto, Legal Bureau, Ministry of Finance, Jakarta Irau: Mohammad Reza Saveji, Ministry of Economic Affairs and Finance, Teheran

Ireland: Dr. Miriam Hederman O'Brien, Trinity College, Dublin Isle of Man: Richard Vanderplank, Cains Advocates & Notaries, Douglas

Israel: Sharon Sachar-Porag, Tel Aviv

Italy: Prof. Avv. Augusto Fantozzi and Andrea Manganelli, Studio Legale Tributario, A. Fantozzi - L. Biscozzi, Rome; Francesco Tundo, Studio Tributario Internationale, Milan

Japan: Akira Akamatsu, Akamatsu Zeirishi Office, Tokyo; Yoichi Ishizuka, Grant Thomton Shinko,

Jersey: J. Paul Frith, Ernst & Young, St. Helier

Jersey: J. Paul Frith, Ernst & Young, St. Heiter
Kazakhstan: Lisa A. Gialdini and Erlan Dosymbekov, Arthur Andersen, Almaty
Kenya: Graham Glenday, Fellow, Harvard Institute for International Development, and Senior Advisor on
Tax Policy, Ministry of Finance, Nairobi
Korea: Sungsik C. Ahn, Young Wha Accounting Corp., Seoul; Yong Kyun Kim, Coopers & Lybrand,
Seoul, J.Y. Lee and R.M. Donaldson, Kim & Chang, Seoul

Kuwait: Abdullah Kh. Al-Ayoub, Kuwait

Latvia: Jevgenijs Ostaskovs, Raimonds Markevics, and Andrejs Birums, State Finance Inspectorate, Riga Lebanon: Fuad S. Kawas, Beirut

Liechtenstein: Reto H. Silvani, Coopers & Lybrand, Liechtenstein

Lithuania: Grazvyda Kveselaitiene, Deputy Head of Revenue Department, Ministry of Finance, Vilnius Luxembourg: Julie Hornsby, Arthur Andersen & Co., Luxembourg

Luxembourg: Julie Hornsby, Arthur Andersen & Co., Luxembourg
The Former Yugoslav Republic of Macedonia: Vanja Mihajlova, Ministry of Finance, Skopje
Madeira: Joac Machado, Ernst & Young, Funchal
Malawi: Clement L. Mononga, Assistant Commissioner of Tax, Blantyre
Malaysia: Prof. bin Bardai Barjoyai, University of Kebangsaan Malaysia, Selangor
Malta: Dr. Antoine Fiott, Zammit Tabona Bonello & Co., and Lecturer in Taxation, Faculty of Law,
University of Malta, Valletta.
Mauritius: Ram L. Roy, Price Waterhouse, Port Louis

Mexico: Luis Manuel Perez de Acha, Mexico City; Manuel F. Solano, Coopers & Lybrand, Mexico City Monaco: Eamon McGregor, Moores Rowland Corporate Services, Monte Carlo Mongolia: Baldangiin Ganhuleg, General Department of State Taxation, Ministry of Finance, Ulaanbaatar Nauru: Peter H. MacSporran, Melbourne

Nepal: Prem Karki, Kathmandu Tax Office, Kathmandu

The Netherlands: Eric van der Stoel, Deloitte & Touche, Tohmatsu International, Rotterdam; Jan A. Ter Wisch, Deloitte & Touche, Tohmatsu International, Rotterdam; Peter Spaans, Coopers & Lybrand, the Hague; Dick Holland, Loyens & Volkmaars, Amsterdam

Netherland Antilles: Dennis Cijntje, KPMG Meijburg & Co., Curaao; Koen Lozie, St. Maarten Int'l Trust Co., St. Maarten

New Zealand: John Shewan, Coopers & Lybrand, Wellington

Nigeria: Elias Aderemi Sulu, Lagos
Northern Mariana Islands: John A. Manglona, Saipan

Norway: Prof. Arvid Aage Skaar, Norwegian School of Management, Sandvika and Wiersholm, Mellbye & Bech, Oslo

Oman: Fudli R. Talyarkhan, Ernst & Young, Muscat Pakistan: Ahmad Khan, Chief of Direct Tax Policy, Central Board of Revenue, Islamabad

Panama: Leroy Watson, Arias, Fabrega & Fabrega, Panama City; Joel R. Medina, Icaza, Gonzalez-Ruiz & Aleman, Panama City

Papus New Guinea: Lutz K. Heim, Ernst & Young, Port Moresby
People's Republic of China: Prof. Chen Xiaohong, Beijing; Wang Xuan Hue, China International Tax
Consultancy Co., Beijing; Arthur Ho, Price Waterhouse, Hong Kong; Professor Jinyan Li, University of
Western Ontario, London, Ontario

Peru: Ismael Noya De La Piedra, Estudio Juris Echecopar Garcia, Lima
The Philippines: Gaudencio Adubo Mendoza Jr., Manila; Cirilo P. Noel, Sycip, Gorres, Velayo & Co.,

Manila
Portugal: Raul Jorge Correia Esteves, Lisbon
Qatar: Finbarr Sexton, Ernst & Young, Doha
Romania: Radu D. Popa, New York University School of Law, New York
Russia: Jeffey Trinklein, Gibson, Duna & Crutcher, Dallas; Jeff Kadet, Arthur Andersen & Co.,
Moscow; Scott C. Antel, Arthur Andersen, Moscow
Saint Kitts-Nevis: Vincent K. Hubbard, Morning Star Holdings, Charlestown
Sierra Leone Shakib N.K. Basma and Berthan Macaulay, Basma & Macaulay, Freetown
Singapore: Michael G. Velten, Baker & McKenzie, Singapore

Slovak Republic: Alzbeta Bobáková, Ernst & Young LLP, Chicago

South Africa: Marius van Blerck, Johannesburg, Kevin Cron, Deneys Reitz Attorneys, Johannesburg Spain: Nicasio del Castillo, Ramón Mullerat & Juan Francisco Rodriguez, Coopers & Lybrand, New York & Madrid; Mariin Echániz, L.C. Rodrigo Abogados, Madrid

Sri Lanka: D.D.M. Waidyasekera, Mt. Lavinia

Taiwan: Keye S. Wu, Baker & McKenzie; Yu Ming-i, Ministry of Finance, Taipei Tanzania: Awadhi Mushi, Dar-es-Salaam

Thailand: Nickane Sucharitaves, SGVN Legal & Tax Consultants Ltd., Bangkok Trinidad & Tobago: Rolston Nelson, Port of Spain

Trunisia: Lassaad M. Bediri, Hamza, Bediri & Co., Legal and Tax Consultants, Tunis
Turkey: Erian Özdemir, Özdemir Financial Consultancy Ltd. Co., Istanbul
Turks & Caicos Islands, British West Indies: Ariel Misick, Misick and Stanbrook, Grand Turk

Uganda: Frederick Ssekandi, Kampala
Ukraine: Rebecca A. Morrison, Ernst & Young, Kiev; George Sharpe, Ernst & Young, Kiev
United Arab Emirates: Nicholas J. Love, Ernst & Young, Abu Dhabi
U.K.: Charles Vitez, Charles Vitez & Co., Middlesex U.S. Virgin Islands: Marjorie Rawis Roberts, Bureau of Internal Revenue, St. Thomas
Uruguay: Alberto Barreix, International Tax Program, Harvard Law School, Cambridge, Massachusetts;
Hugo Valerno, Head, Department of Tax Policy, Internal Revenue Service, Montevideo
Vanuatu: Bill L. Hawkes, KPMG, Port Vila

Venoruela: McLorm Caplan, Baker & McKenzie, Caracas
Western Samoa: Maiava V.R. Petern, Kamu & Petern, Apia
Yugodavia: Dr. Zeljko Bogetic, The World Bank, Washington D.C.
Zambia: P C Luhanga, KPMG Peat Marwick, Lusaka
Zimbabwe: Prof. Ben Hlatshwayo, University of Zimbabwe, Harare.

## Tax Policy Forum





## Puerto Rico and Section 936: A Costly Dependence

by J. Tomas Hexner and Glenn P. Jenkins

J. Tomas Hexner is President of the International Institute for Advanced Studies. Glenn P. Jenkins is Director of the International Tax Program at the Harvard Law School and Institute Fellow of the Harvard Institute for International Development, Harvard University.

The authors would like to thank Elizabeth Chant, Harry Hjardemaal, and Jeffrey Hall for assistance with research and editing in the preparation of this report. The comments of Richard Ainsworth and Chung-Hee Lee on an earlier draft of this paper are greatly appreciated. All errors and omissions are the responsibility of the authors alone.

#### I. Introduction

For over 70 years, U.S. corporations have been granted tax incentives to operate in U.S. territorial possessions, most notably in Puerto Rico. The purpose in so benefiting what have become known as "possessions corporations" is to attract U.S. capital to these developing territories, with the goal of creating jobs.

At the outset, this approach—as expressed first in section 262 of the Revenue Act of 1921, and ultimately in section 936 of the Internal Revenue Code—was successful. In Puerto Rico during the 1950s and 1960s, it spurred the island's industrialization, infrastructure development, and

the attendant growth in employment and gross national product (GNP). By the mid-1970s, however, the job-creation benefits of section 936 took a backseat to the tax planning that brought great financial gain to only a few U.S. companies, substantial cost to the U.S. Treasury, and a competitive disadvantage to "native" Puerto Rican enterprises.

This problem persists in large part because section 936 can have an immense impact on after-tax U.S. profits: possessions corporations receive full credit against U.S. taxes owed on the net income earned in a possession, regardless of whether that income is generated by the use of tangible property and labor within the pos-

session or is attributable to the use of intangible property transferred to the possession corporation. Companies have been quick to see the benefit in transferring intangible assets and their related income streams to their possessions-based operations. Therefore, to the extent that corporations are able to claim tax credits for income sourced in the possessions that has been generated by properties temporarily located there but for which no real investments have been made, the cost of section 936 to the U.S. Treasury has been wholly unrelated to the intended development benefits that underlay adoption of section 936 in the first instance.

Some analysts, considering the section 936 problem solely from the perspective of the revenue loss caused by the artificial transfers of intangible assets to possessions

<sup>&</sup>lt;sup>1</sup>These territorial possessions now include Puerto Rico, the U.S. Virgin Islands, the Commonwealth of the Northern Marianas, the Federated States of Micronesia, the Marshall Islands, American Samoa, and Guam. The Philippines was considered a U.S. possession until 1946, when it was given its independence.

<sup>&</sup>lt;sup>2</sup>A possessions corporation is a U.S. corporation, which is commonly the subsidiary of a U.S. parent corporation doing business in a U.S. territorial possession, and which otherwise qualifies for the special tax credit afforded to such corporations under section 936 of the Internal Revenue Code (IRC).

corporations, have sought remedies by linking section 936 with section 482 of the code. Other analysts, considering the 936 problem solely from the perspective of the lack of real, development-based investments in the possessions, have sought remedies in credit limitations based on measures of real investments in the possessions themselves. Recent additions to the Internal Revenue Code reflect solutions from both perspectives.

This paper argues that neither solution offers much hope of real success, and that the current mix of solutions, although well intended, most likely will have a serious negative impact on "native" possession enterprises and the revenues these enterprises contribute to possession treasuries.

It is time to admit that the attractiveness of section 936 as a tax scheme has come to far outweigh its role as an employment-producing incentive. The companies benefiting most from the credit have been capital-intensive firms such as pharmaceutical companies. Those benefiting least have been labor-intensive industries such as apparel manufacturers.

For example, in Puerto Rico during the 1980s, the pharmaceutical industry received about 50 percent of the total tax benefits from section 936 while providing only 15-18 percent of the section 936 jobs. In 1989, the latest year for which aggregate data are available, this translated into the pharmaceutical industry receiving \$1.2 billion of all section 936 credits, while employing only about 18,000 of the 106,000 workers in section 936 firms. The average cost to the U.S. Treasury for each Puerto Rican job in the pharmaceutical industry that year was \$66,081, while the average compensation was \$30,447. Thus, for each dollar of employee compensation, pharmaceuticals received \$2.17 in tax benefits from the U.S. Treasury.<sup>3</sup>

The total cost of the section 936 tax credit to the U.S. Treasury in 1989 was approximately \$2.5 billion. The present value of its cumulative cost during 1973-89 is approximately \$52 billion. The Treasury Department's Office of Tax Analysis projected that the costs of section 936, were it not revised, would continue rising at some 10 percent annually, while the Congressional Budget Office

The cost of section 936 to the U.S. Treasury has been wholly unrelated to the intended development benefits that underlay adoption of section 936 in the first instance.

calculated that the incentive scheme would bring losses of \$15 billion in potential tax revenues during 1993-97.7

These conditions made section 936 a logical target for deficit-reduction legislation in President Clinton's 1993 budget. The revised section 936 provisions restructure and reduce the tax credit effective December 31, 1993. In particular, Congress has legislated a connection between the tax credit and employment and investment growth in the possessions.

Although reform is desirable, this paper argues on a historical basis that section 936 should not merely be fixed, and indeed, that it cannot be fixed. Our contention is that section 936 has:

 essentially operated as a costly tax benefit to a few corporations;

- resulted, through links with the Caribbean Basin Initiative, in substantial gains for possessions corporations with little corresponding boost in regional exports;
- created a tax-subsidy-oriented development strategy, which for the past 20 years has been a principal cause of stagnation in the Puerto Rican economy; and
- has generated, understandably, a powerful lobbying effort to perpetuate the 936 corporate financial benefits by delivering the message that the Puerto Rican economy would falter without the investment stimulus of 936.8

We find further that the revisions to section 936, as provided in President Clinton's 1993 budget, do not address these shortcomings. Moreover, like earlier attempts to fix the tax benefit, they promise a result that is inferior to the possibilities of abandoning the policy entirely.

<sup>&</sup>lt;sup>3</sup>J. Bradford, "U.S. Possessions Corporations Returns, 1989," U.S. Department of the Treasury, Office of Tax Analysis, p. 103.

 $<sup>^4</sup>Id.$ 

<sup>&</sup>lt;sup>5</sup>U.S. Department of the Treasury, "The Operations and Effect of the Possessions Corporation System of Taxation, Sixth Report," 1989, Table 4-11. Figures for 1984 and 1986 were imputed by taking the mean between available data for 1985 and 1987. The discount rate used was 8 percent.

<sup>&</sup>lt;sup>6</sup>P. Morrison, "Testimony before the Committee on Finance, United States Senate," April 26, 1990, p. 2.

<sup>&</sup>lt;sup>7</sup>U.S. General Accounting Office, "Pharmaceutical Industry: Tax Benefits of Operating in Puerto Rico," Briefing Report to the Chairman, Special Committee on Aging, U.S. Senate, May 1992, p. 1.

<sup>&</sup>lt;sup>8</sup>See J.T. Hexner, G. Jenkins, H.F. Ladd, and K.R. LaMotte, Puerto Rican Statehood: A Precondition to Sound Economic Growth, November 1993. This report shows that section 936 acts as an unsustainable crutch in the Puerto Rican economy and, in so doing, creates significant market distortions, thereby impeding the economic development of the island.

Section II of this article reviews the historical background of section 936, and section III examines its mechanics and looks at the technical aspects of related legislation. In particular, this article explores the relationship between section 936 and regulations issued to limit transfer pricing abuses. Section IV analyzes the impact of section 936 on Puerto Rico. Section V explores the legal relationship between section 936 and the Caribbean Basin Initiative—an act that will prolong the tax incentive. Section VI outlines the elements of the most recent attempts to reform the tax credit. Section VII concludes that section 936 cannot and should not be fixed because, as a development strategy, it is expensive and ineffective—expensive to U.S. taxpayers and ineffective in stimulating the growth of the Puerto Rican economy.

#### II. The Legislative History of Section 936

#### A. Background

Since the Revenue Act of 1921 (with its section 262, the predecessor of section 936), the U.S. government has provided a tax incentive for U.S. corporations operating in its territorial possessions. The original goal was to help American corporations compete with foreign firms in the Philippines. The original goal was to help American corporations compete with foreign firms in the Philippines.

Section 262 exempted qualified U.S. corporations from taxes on all income derived from sources outside the United States. To qualify, a corporation had to derive 80 percent or more of its gross income from its operations in U.S. possessions, and 50 percent or more of its gross income from active trade or business in the possessions.11 These gross income tests had to be met on an aggregate basis for the year of the exemption and for the two preceding tax years if the corporation had conducted a trade or business in a possession during that period. Under the 1921 act, dividends paid by the possessions corporation to corporate shareholders were fully taxable. In the Revenue Act of 1935, however, this policy was abandoned. Moreover, amounts received upon liquidation were made tax-exempt.<sup>12</sup>

In 1948, by coupling these U.S. tax incentives with various local tax incentives, Puerto Rico initiated a more aggressive program to attract major capital investment. This program, known as

The original goal of section 936 was to help American corporations compete with foreign firms in the Philippines.

"Operation Bootstrap,"13 attracted a surge of U.S. corporations, particularly in labor-intensive industries. From 1948 to 1972, Puerto Rico's real GNP grew at an average annual rate of 6 percent (compared to a rate of 3.7 percent for the United States).14 At the same time, the island's economy shifted from its traditional agricultural base to manufacturing, where employment increased from 55,000 in 1950 to 142,000 in 1972.15 Indeed, the program was so successful that during the 1950s and 1960s, Puerto Rico was dubbed the "economic miracle" of the Caribbean.16

After this boom, however, the Puerto Rican economy stagnated. And while the section 936 lobby has attempted to maintain and disseminate the historical boom illusion, the annual rate of physical investment declined by nearly 30 percent between 1973 and 1978, from \$1.5 billion to \$1.1 billion.<sup>17</sup> In the next five years, from 1978

to 1983, new physical investment fell another 35 percent, from \$1.1 billion to \$0.7 billion. 18 Private investment in plant and equipment also fell steadily from 10.3 percent of GNP in 1973 to 4.6 percent in 1983. 19

By the mid-1970s, the possessions tax benefit began to be criticized as an insufficient stimulus for employment-producing investments in Puerto Rico and the other possessions. Later, during the 1980s, other criticism emerged to the effect that, because the tax incentive provided that liquidation receipts were taxexempt, possessions corporations were accumulating and investing earnings in the Eurodollar market and other foreign markets for long periods before liquidating and repatriating these earnings

<sup>&</sup>lt;sup>9</sup>Revenue Act of 1921, ch. 136, section 262, 42 Stat. 227, 314.

<sup>&</sup>lt;sup>10</sup>The vast majority of section 936 companies conduct business in Puerto Rico. In 1987, nearly 97 percent of all U.S. possessions corporations operated in Puerto Rico and over 99 percent of the total section 936 credit was claimed by companies with operations in Puerto Rico. See J. Bradford, "U.S. Possessions Corporations Returns," p. 51. Consequently, this article will focus on the operation of section 936 in Puerto Rico.

<sup>&</sup>lt;sup>11</sup>U.S. Department of the Treasury, "Sixth Report," p. 5.

<sup>&</sup>lt;sup>12</sup>Revenue Act of 1935, ch. 829, sec. 112(b)(6), 49 Stat. 1014, 1020.

<sup>&</sup>lt;sup>13</sup>The Operation Bootstrap program was conceived by Puerto Rican Governor Luis Munoz Marin and promised U.S. corporations "cheap labor, exemptions from island taxes for up to 25 years (along with total exemption from U.S. federal corporate and private income taxes), and assistance in the building of plants." Tansill, "Puerto Rico: Independence or Statehood?" Revista del Colegio de Abogados de Puerto Rico 41 (1980): 93.

<sup>&</sup>lt;sup>14</sup>U.S. Department of the Treasury, "Sixth Report," pp. 17, 19.

<sup>&</sup>lt;sup>15</sup>Id., p. 17.

<sup>&</sup>lt;sup>16</sup>Id.

<sup>&</sup>lt;sup>17</sup>*Id.*, p. 24.

<sup>&</sup>lt;sup>18</sup>Id.

<sup>&</sup>lt;sup>19</sup>Id. p. 17.

tax-free to the United States.<sup>20</sup> By the mid-1980s, opponents of the tax benefit further argued that its cost in foregone tax revenue contradicted deficit-reduction efforts by the U.S. Treasury.<sup>21</sup>

Those favoring a continuation of the tax exemption countered that the incentives were needed to offset the costs of federally imposed requirements in the possessions. U.S. law, for example, set minimum wages and mandated the use of U.S. flag vessels to transport goods to the mainland. This was said to disadvantage Puerto Rico, and other U.S. possessions generally, in competition with other developing countries for U.S. investment.<sup>22</sup>

Congress responded to the early criticisms by creating a new section 936 of the Internal Revenue Code in the Tax Reform Act of 1976.<sup>23</sup> Congress stated that it sought to

... assist the U.S. possessions in obtaining employment-producing investments by U.S. corporations, while at the same time encouraging those corporations to bring back to the United States the earnings from these investments to the extent they cannot be reinvested productively in the possession.<sup>24</sup>

The essence of the 1976 legislation remained intact and continued to apply until December 31, 1993. Its unsatisfactory performance, with respect to the goals of Congress, is, however, broadly apparent. The benefit to muchneeded employment in Puerto Rico continues to be low (the unemployment rate in Puerto Rico is now 18.1 percent) relative to its mounting cost (\$2.5 billion in 1989) to the U.S. Treasury. This result has occurred because the legislation supported (and continued to support until December 31, 1993) possessions corporations and their affiliates in the exploitation of transfer pricing.

#### B. Combating Transfer Pricing Abuses

Before 1982, there were no explicit statutory guidelines on transfer pricing. <sup>25</sup> This statutory silence provided possessions corporations with tacit permission to minimize their tax liability by shifting the taxable income attributable to property transferred from U.S. affiliates. A U.S. pharmaceutical company, for example, might develop a patentable drug

transferred intangibles would be allocated to the U.S. parent.<sup>27</sup> Section 936(h) was revised again in 1986 to coordinate with section 482 provisions, which address transfer pricing in general. And, as recently as January 1993, Congress once again revised the regulations when it issued new temporary section 482 regulations, which refer to section 936(h).

The 1976 legislation supported (until December 31, 1993) possessions corporations and their affiliates in the exploitation of transfer pricing.

in its U.S. laboratory and receive deductions on its U.S. federal income tax obligations for the research and development costs it incurred. The company would then transfer the patent to its wholly owned possessions corporation, which would produce the patented drug and would claim the resulting income as possession-source income. As a result, the corporate group would owe little or no income tax, either in the United States or in Puerto Rico, for producing this drug.<sup>26</sup>

Congress and the Treasury have repeatedly reacted to this problem but have met with limited success. Congress enacted the Tax Equity and Fiscal Responsibility Act of 1982 (TEFRA), which added a new section 936(h) to the Internal Revenue Code to ensure that a sufficient percentage of income generated by such

<sup>24</sup>U.S. House of Representatives, "Report on H.R. 10612," p. 255; and U.S. Senate, "Report on H.R. 10612," p. 279.

<sup>25</sup>U.S. Department of the Treasury, "Sixth Report," p. 8. U.S. corporations operating in the possessions usually show profits in two ways. First, they earn profits from real investment in plant and equipment in Puerto Rico, Second, they are sometimes able to increase the amount of accounting profits reported in the possessions without any new physical investment by allocating to a possessions corporation income from intangibles (such as patents, trademarks, and trade names) that had typically been developed and paid for by an affiliated U.S. corporation and subsequently were transferred to the possessions corporation at a transfer price that does not reflect the market cost or the costs of development.

<sup>26</sup>Id.; see also U.S. General Accounting Office, "Pharmaceutical Industry," p. 2. The U.S. Treasury took the opposite position, however, and argued that income obtained from drug sales in these transactions should be allocated to the U.S. parent and was subject to federal taxation. This issue resulted in lengthy litigation. See, for example, Eli Lilly and Co. v. Comm'r., 84 T.C. 996 (1985) and G.D. Searle & Co. v. Comm'r., 88 T.C. 252 (1987).

 $^{27}$ U.S. Department of the Treasury, "Sixth Report," p. 8.

<sup>&</sup>lt;sup>20</sup>Id., p. 7.

<sup>&</sup>lt;sup>21</sup>*Id.*, p. 6.

 $<sup>^{22}</sup>Id.$ 

<sup>&</sup>lt;sup>23</sup>See generally U.S. House of Representatives, "Report of the Committee on Ways and Means, U.S. House of Representatives, on H.R. 10612," Report No. 94-658, November 12, 1975; and U.S. Senate, "Report of the Committee on Finance, U.S. Senate, on H.R. 10612," Report No. 94-938, June 10, 1976.

#### C. Section 936 Eligibility and Links to Investment and Employment

Congress has undertaken a parallel effort to tighten the eligibility requirements for the tax exemption. It has repeatedly revised the gross income test (the minimum percentage of a section 936 firm's income that must be earned from the active conduct of trade or business in the possessions to qualify for the tax credit). Revisions to section 936 in 1976 set the minimum at 50 percent the same figure required under the antecedent legislation. The 1982 revision increased the minimum to 65 percent, and in 1986 it was raised again to 75 percent.28 Hence, a section 936 firm may now derive no more than 25 percent of its gross income from passive investments.

Stipulations in the Clinton administration budget adopted in 1993 represent the most recent attempt to make section 936 "work." These provisions, which came into effect on December 31, 1993, aim to reduce the tax credit while strengthening its link to investment, employment, and wage growth in the possessions.

The following sections evaluate the specifics of the evolving section 936 legislation and related provisions.

#### III. The Mechanics of Section 936 and Related Legislation

#### A. The Section 936 Tax Credit

Section 936 grants to subsidiaries of U.S. corporations operating in the possessions a tax credit<sup>29</sup> equal to the U.S. federal income tax liability from such operations.<sup>30</sup> This credit is based on the taxable income derived from: (1) trade or business within the possession,<sup>31</sup> (2) the sale or exchange of substantially all of the assets used by the subsidiary in this trade or business,<sup>32</sup> and (3) "qualified possession source investment income, (QPSII)" (i.e., passive income resulting from in-

vestment in the possessions of the exempted profits). $^{33}$ 

The credit is available to any U.S. corporation that during the three years prior to the close of the tax year (or for such part of such period immediately preceding the close of the tax year as may be applicable) earned 80 percent or more of its gross income from possession sources,<sup>34</sup> and earned 75 percent or more of its gross income from the active conduct of trade or business within the possessions.<sup>35</sup>

A section 936 firm may now derive no more than 25 percent of its gross income from passive investments.

U.S. parent corporations are eligible for a dividends-received deduction on dividends received from a possessions corporation.<sup>36</sup> If the possessions corporation is a wholly owned subsidiary—as most of them are—the deduction equals 100 percent of the dividend.37 Such a possessions corporation can therefore repatriate to its U.S. parent, free of any U.S. federal income tax liability, its income earned in the possessions. Possessions governments may, however, impose their own taxes on earnings of the possessions corporations. This can include, as is the case for Puerto Rico, a tollgate tax on the repatriated earnings.38

Gross income received on the mainland by a possessions corporation is only considered possession-source income if it is derived from trade or business with unaffiliated parties.<sup>39</sup> If a U.S. corporation deposits payments into a

bank account on the mainland of a possessions corporation subsidiary as payment for goods manufactured by that subsidiary in Puerto Rico, the payment will not be considered possession-source income of the subsidiary.<sup>40</sup> The subsidiary must receive payment in Puerto Rico for goods and services in order for the payment to be considered possession-source income that qualifies for the section 936 tax credit.<sup>41</sup>

<sup>29</sup>The dollar amount of the section 936 credit is determined as follows:

Tax Credit =
Taxable Business and
Investment Income
From Sources Within Puerto Rico
Worldwid Taxable Income

Worldwide Taxable Income of Possessions Corporation

x U.S. Tax

See R.J. Boles, "Tax Incentives for Doing Business in Puerto Rico," International Lawyer 22, no. 1 (Spring 1988): 123, which explains the basis for this calculation.

<sup>30</sup>IRC section 936(a)(1) (1989).

<sup>&</sup>lt;sup>28</sup>The 1986 act also expanded the range of types of investment income that qualify for the tax exemption. Income from deposits in Puerto Rican financial institutions that are used to finance development projects in Caribbean Basin Initiative countries now qualify.

<sup>&</sup>lt;sup>31</sup>IRC section 936(a)(1)(A)(i).

<sup>32</sup>IRC section 936(a)(1)(A)(ii).

 $<sup>^{33}</sup>$ IRC section 936(a)(1)(B). The operation of the qualified possession-source investment income provision of section 936 will be discussed in Section V.

 $<sup>^{34}</sup>$ IRC section 936(a)(2)(A).

 $<sup>^{35}</sup>$ IRC section 936(a)(2)(B).

 $<sup>^{36}</sup>$ IRC section 243(b)(1)(C).

 $<sup>^{37}</sup> U.S.$  Department of the Treasury, "Sixth Report," p. 7.

<sup>&</sup>lt;sup>38</sup>Puerto Rican tax laws applicable to U.S. possessions corporations are discussed in the next part of this section.

<sup>&</sup>lt;sup>39</sup>IRC section 936(b).

<sup>&</sup>lt;sup>40</sup>Pacific Basin Mfg. & Trade Co. v. Comm'r., 716 F.2d 638 (9th Cir. 1983); Rev. Rul. 79-168, 1979-2 C.B. 283.

<sup>&</sup>lt;sup>41</sup>Nevertheless, the standard foreign tax credit may be claimed for foreign taxes paid or accrued on income that does not qualify for the section 936 credit. See U.S. Department of the Treasury, "Sixth Report," p. 7.

A possessions corporation may not join in a consolidated return with its parent or any affiliated corporations, even in a year in which it fails to satisfy either the 80-percent possessions-source test or the 75-percent active trade or business test. 42 Hence, operating losses incurred by a possessions corporation may not offset the taxable income of the parent or an affiliated corporation. This means that a subsidiary engaged in trade or business in a possession ordinarily will not elect to file under section 936 until it is no longer incurring start-up losses.43

The section 936 tax credit is not available for use against the environmental tax,44 the tax on accumulated earnings,45 the personal holding company tax,46 or taxes arising out of recoveries of foreign expropriation losses. 47 For purposes of the accumulated earnings tax, the accumulated taxable income of a possessions corporation does not include taxable income eligible for the section 936 credit.48 The credit is also unavailable to a corporation for any tax year in which that corporation is a domestic international sales corporation (DISC) or former DISC.49 or for any tax year in which it owns stock in a DISC or former DISC.50 or in a foreign sales corporation (FSC) or former FSC.51

A possessions corporation may elect to use section 936 by filing Treasury Form 5712. For the first tax year in which a possessions corporation applies for the section 936 credit, the form must be submitted on or before the date on which the federal income tax return is filed.<sup>52</sup> An election to use the credit may not be revoked for a period of 10 years without consent from the secretary of the Treasury.<sup>53</sup>

#### B. Complementary Puerto Rican Tax Incentives

In addition to the tax credit provided under section 936, the Puerto Rican government has, since 1948, provided its own complementary tax incentives for manufacturing and other specified business activities. Puerto Rico currently grants partial exemptions (of 90 percent) from income tax and other taxes to approved businesses for specified periods of time, usually from 10 to 25 years. <sup>54</sup> A business is generally eligible for an exemption if it is producing on a commercial scale in Puerto Rico a "designated service unit" or a manufactured product not produced in Puerto Rico before January 1, 1947. <sup>56</sup>

Companies that meet that criterion are entitled to a 90-percent income tax exemption, for a period that varies according to the level of business activity in the area where the business is located:<sup>57</sup>

Location	Duration of Exemption
High Development Zone	10 years
Intermediate Development Zone	15 years
Low Development Zone	20 years
Vieques or Culebra	25 years

Qualified manufacturers also receive partial exemptions (up to 90 percent) from property taxes on the personal or real property that generates the exempted income. Moreover, a manufacturing company with gross income of less than \$500,000 in any year and with average employment that year of at least 15 persons receives a 100-percent deduction of its first \$100,000 of income. A 60-percent exemption from municipal license (gross receipts) taxes is also granted. Businesses that qualify for these exemptions are subject to a special surcharge equal to the lesser of 0.075 percent of sales or 0.5 percent of net income if their income is in excess of \$100,000 in a tax year.58

A tollgate tax of up to 10 percent may be imposed on earnings repatriated to the United States

or to a foreign country. The rate depends on the amount and the length of time that these earnings were invested in Puerto Rico prior to their repatriation. Holding earnings in certain designated investments in Puerto Rico (such as Puerto Rican bonds, bank savings certificates, participation in construction loans, or investment in the company's own additional plant and equipment) for five or more years will decrease the tollgate tax rate by 1 percentage point for each year that the investment is maintained. Thus, earnings invested in these instruments for six years will result in a 4-percent tollgate tax rate when the earnings are repatriated. If

<sup>&</sup>lt;sup>42</sup>R.J. Boles, "Tax Incentives," p. 125.

<sup>&</sup>lt;sup>43</sup>To the extent that any losses prior to electing section 936 status have been used beneficially to offset the U.S.-source income of an affiliated group, the possessions corporation will ultimately be required to "recapture" such losses by treating them as U.S.-source income under the overall foreign loss recapture rules of IRC section 904(f).

<sup>&</sup>lt;sup>44</sup>IRC section 936(a)(3)(A); IRC section 59A.

<sup>&</sup>lt;sup>45</sup>IRC section 936(a)(3)(B); IRC section 531.

 $<sup>^{46}</sup>$ IRC section 936(a)(3)(C); IRC section 541.

<sup>&</sup>lt;sup>47</sup>IRC section 936(a)(3)(D); IRC section

<sup>&</sup>lt;sup>48</sup>IRC section 936(g).

<sup>&</sup>lt;sup>49</sup>IRC section 936(f)(1).

<sup>&</sup>lt;sup>50</sup>IRC section 936(f)(2)(A).

<sup>&</sup>lt;sup>51</sup>IRC section 936(f)(2)(B).

 $<sup>^{52}</sup>IRC$  section 936(e)(1); R.J. Boles, "Tax Incentives," p. 123.

 $<sup>^{53}</sup>$ IRC section 936(e)(2).

<sup>&</sup>lt;sup>54</sup>Puerto Rico Tax Incentives Act, section 3, 13 L.P.R.A. section 256b(a) (Supp. 1988) (approved Jan. 24, 1987).

<sup>&</sup>lt;sup>55</sup>13 L.P.R.A. section 255a(d)(4). The term "designated service unit" applies to certain service production activities such as distribution, investment banking, public relations, publicity, consulting, and computer services.

<sup>&</sup>lt;sup>56</sup>13 L.P.R.A. section 256a(d)(1).

<sup>&</sup>lt;sup>57</sup>13 L.P.R.A. section 256(b)(d).

<sup>&</sup>lt;sup>58</sup>13 L.P.R.A. section 256b(a).

**XYZ Corporation** 

Hypothetical Subsidiary Operation in the U.S. and Puerto Rico Pharmaceutical Industry

	Manufacturing United States	Plant Location Puerto Rico
Sales	\$150,000,000	\$150,000,000
Income Before Taxes	50,000,000	50,000,000
Effective Corporate Tax Rate	35%	4.5%
Income Taxes	17,500,000	2,228,750
Special Surtax Rate	_	0.075%
Special Surtaxes	0	112,500
Tollgate Tax <sup>1</sup>	0	4,765,875
Net Income After Tax	32,500,000	42,892,875
Tax Savings <sup>2</sup>	0	10,392,875

<sup>&</sup>lt;sup>1</sup>The 10-percent tollgate tax applied assumes immediate repatriation of earnings.

the amount invested is at least 50 percent of the income of the exempted business for a given year, then all of that year's earnings will qualify for the reduced tollgate tax rate. The 50 percent (or less) of net income not invested can be repatriated immediately at the reduced rate. At the end of the investment period, the invested funds also can be repatriated at the reduced rate. <sup>59</sup>

Example No. 1 illustrates how these rules operate. 60 It shows that if located in Puerto Rico, 90 percent of the income of the subsidiary of XYZ company would be exempt from local income taxes. The remaining 10 percent would be taxed at a rate of 45 percent. Also, the 0.075-percent surtax on sales and the tollgate tax on repatriated earnings would apply. Consequently, the tax would be \$2,250,000 (10 percent of the \$50,000,000 in income taxed at a rate of 45 percent) less \$21,250.61

This amounts to a total of \$7.11 million owed to the government of Puerto Rico on income of \$50 million, compared to an estimated

\$17.5 million that would be owed on similar income derived from mainland operations. The effective tax rate for this company thus is only 14.22 percent (the sum of \$7.11 million in income tax, surtax, and tollgate tax divided by \$50 million in income), compared to the 35-percent maximum corporate tax rate the corporation would face on similar operations in the United States. As this example shows, a U.S. company that operates in Puerto Rico under section 936 stands to reap a substantial increase in net aftertax profits through the drastic reduction in tax liability available on the island.

## C. The TEFRA Amendments to Section 936

Since 1982, the Tax Equity and Fiscal Responsibility Act (TEFRA) has provided statutory rules for the allocation to a possessions corporation of income from intangibles that were developed or purchased by its affiliated corporations. The act is one in a series of attempts by the U.S. Con-

gress to stem transfer pricing abuse.

TEFRA added a new section 936(h) to the Internal Revenue Code. This section provides that income from intangible property that is not owned by a possessions corporation is not eligible for the section 936 tax credit. Rather, it is generally taxable to the U.S. shareholders of the possessions corporation. TEFRA provides further that a possessions corporation and its affiliates may elect out of this general rule under either a "cost-sharing" option or a "50/50 profit-split" option. 62

These two options provide methods by which a possessions corporation may claim an appropriate portion of the income from intangible property that is transferred from its affiliates. If the possessions corporation does not elect either method, it must compute its income from intangible property based on a reasonable profit on the costs that are attributable to such income. <sup>63</sup>

The cost-sharing and profitsplit options apply only to "possession products," products produced wholly or partially by a possessions corporation.<sup>64</sup> The possessions corporation must elect to treat all products in the same pro-

 $<sup>^2</sup>$ Tax savings is the difference in potential income tax obligations between the U.S. and Puerto Rico. In this example, the tax savings equal \$17.5 million minus \$7.11 million.

<sup>&</sup>lt;sup>59</sup>13 L.P.R.A. section 256c(b).

<sup>&</sup>lt;sup>60</sup>Based on examples given by the U.S. Department of the Treasury, Internal Revenue Service, in C.F.R.

<sup>&</sup>lt;sup>61</sup>See R.J. Boles, "Tax Incentives," Appendix B, p. 142, which explains the basis for this calculation.

<sup>62</sup>IRC section 936(h)(5).

<sup>&</sup>lt;sup>63</sup>R.J. Boles, "Tax Incentives," p. 129.

<sup>&</sup>lt;sup>64</sup>The regulations under IRC section 936(h) provide a flexible definition of the term "possession product." The term includes any item of property that is the result of a production process, including components and so-called "end-product forms." End-product forms are products that are treated as not including certain other components for purposes of meeting the business-presence test and for the computation of the amount of income derived from the possession product.

duct area (defined by reference to three-digit classification using the Standard Industrial Classification (SIC) code) in a like manner. <sup>65</sup> If a corporation elects one of these options, it may, however, make a different election for export and domestic sales. <sup>66</sup> To be eligible to use either the cost-sharing or profit-split option, a possessions corporation must have a "significant business presence" with respect to a particular product in a possession. This requires meeting one of two tests:

#### 25-Percent Value Added

Test: The possessions corporation must show that it incurred production costs<sup>67</sup> with respect to the product that are not less than 25 percent of the difference of (1) gross receipts from sales or other disposition of the product to unrelated parties by the possessions corporation or its affiliates less (2) direct costs of materials purchased by the possessions corporation or its affiliates from unrelated parties in connection with the manufacture of that product.<sup>68</sup>

65-Percent Labor Test: Alternatively, the possessions corporation must show that it incurred at least 65 percent of the total direct labor costs<sup>69</sup> of the possessions corporation and its affiliates in producing the product or service during the tax year. The 65 percent refers to compensation for labor services performed in the possession.<sup>70</sup>

Start-up operations of new 936 corporations and new possession products of existing 936 corporations can meet the "significant business presence" requirement by satisfying a lower threshold of value added or labor cost than the percentages referred above. For such operations, a transition period is provided, as follows:

	Year 1	Year 2	Year 3
Value Added Test	10%	15%	20%
Labor Test	35%	45%	55%

#### 1. The Cost-Sharing Option

Under the cost-sharing option, a possessions corporation is reguired to make a payment to its U.S. parent for 110 percent of its share of the cost (if any) of product-area research that is paid or accrued by the affiliated group during the tax year.71 "Product area research" costs include research, development, and experimental costs, losses, expenses, and other related deductions, including amounts paid for the use of, or right to use, a patent, invention, formula, process, design, pattern, or know-how (or the amount paid for the acquisition of any of these) that are allocable to the same product area as that in which the possessions corporation conducts its activities. Also included is a pro rata portion of any costs, expenses, and other deductions that cannot definitely be allocated to a particular product area.72

The payment required of the possessions corporation is therefore 110 percent of a portion of the year's research expenditures of the affiliated group in the product area in which the possession product falls.73 This portion is defined as the ratio of (1) third-party sales of the possession product made by the affiliated group to (2) thirdparty sales of all products in the product area made by the affiliated group.74 The cost-sharing payment is determined separately for each product by using the following formula:

> Sales to Unrelated Persons of Possession Product

Total Sales of Products in 3-digit SIC Code

110% of Worldwide Product-Area Research

Cost-Sharing Payment

A possessions corporation may credit its payments under costsharing arrangements with unrelated persons against its share of the cost of product area research paid or accrued by the affiliated group. On the other hand, amounts paid to, or on behalf of, related persons and amounts paid under any sharing agreements with related persons may not be credited against the possessions corporation's cost-sharing payment for the tax year.<sup>75</sup>

Accordingly, a possessions corporation electing the cost-sharing payment method is treated as the owner of the manufacturing intangibles (but not marketing intan-

<sup>&</sup>lt;sup>65</sup>IRC section 936(h)(5)(C).

<sup>66</sup> IRC section 936(h)(5).

<sup>&</sup>lt;sup>67</sup>For purposes of the value added test, the term "production costs" has the same meaning as in 26 C.F.R. section 1.471-11(b) except that the term does not include direct material costs and interest. Thus, production costs include direct labor costs and fixed and variable indirect production costs (other than interest). Fixed indirect production costs may include, among other costs, rent, and property taxes on buildings and machinery incident to and necessary for manufacturing operations and processes. Variable indirect production costs may include, among other costs, indirect materials, factory janitorial supplies, and utilities. See 26 C.F.R. section 1.471-11(b).

<sup>&</sup>lt;sup>68</sup>IRC section 936(h)(5)(B).

<sup>&</sup>lt;sup>69</sup>Direct labor costs include the cost of labor that can be identified or associated with particular units or groups of units of a specific product. The elements of direct labor include such items as basic compensation, overtime pay, vacation and holiday pay, sick leave pay, shift differential, payroll taxes, and payments to a supplemental unemployment benefit plan paid or incurred on behalf of employees engaged in direct labor. IRC section 936(h)(5)(B).

<sup>&</sup>lt;sup>70</sup>IRC section 936(h)(5)(B).

 $<sup>^{71}</sup>$ IRC section 936(h)(5)(C)(i)(I).

<sup>&</sup>lt;sup>72</sup>IRC section 482 is to be applied if no intangible property is related to a product produced in whole or in part by a possessions corporation (discussed in Section VI).

<sup>&</sup>lt;sup>73</sup>IRC section 936(h)(5)(C).

<sup>&</sup>lt;sup>74</sup>U.S. Department of the Treasury, "Sixth Report," p. 10.

<sup>&</sup>lt;sup>75</sup>Treas. reg. section 1.936-6.

XYZ is a possessions corporation engaged in the manufacture and sale of four products (A, B, C, and D), all of which are classified under the same three-digit SIC code. XYZ sells its production to a U.S. affiliate, P, which resells it to unrelated parties in the United States. P's third-party sales of each of these products produced in whole or in part by XYZ are \$2,000,000 per product, or a total of \$8,000,000 for A, B, C, and D. P's other sales of products in the same SIC code are \$8,000,000. The worldwide product-area research of the affiliated group is \$500,000. XYZ must compute its cost-sharing amount for each individual product A, B, C, and D as follows:

Sales to Unrelated Persons		
of Possession Product	x 110% of Worldwide =	Cost-Sharing
Total Sales of Products	Product Area Research	Payment
in 3-digit SIC Code		
		AAA 550
<u>\$2,000,000</u>	x \$550,000 =	\$68,750
\$16,000,000		

gibles) associated with the possession product. $^{76}$ 

By virtue of a 1986 amendment to section 936(h)(5), the payment made under any cost-sharing option cannot be less than what would be required under section 367(d) or section 482 of the Internal Revenue Code if the electing corporation were a foreign corporation.77 Section 367(d) and section 482 provisions essentially authorize the IRS to reallocate gross income and deductions between affiliated businesses. Such a reallocation is performed when the IRS, following specific guidelines, questions transfer prices and determines that a reallocation is required to stem tax evasion or to reflect income clearly.

Example Nos. 2 and 3 show how to determine the amount of the cost-sharing payment. They are based on examples given by the Treasury Department.<sup>78</sup>

Example No. 2 shows that the amount of the cost-sharing payment would be \$68,750. If, however, XYZ also received \$10,000 in royalty income from an unrelated person for the licensing of certain manufacturing intangible property rights, the amount of the product area research (\$500,000) would be reduced by that amount, to \$490,000.

Example No. 3 shows that a payment by the possessions corporation to an unrelated party under a cost-sharing arrangement will serve to reduce the cost-sharing payment, in this case by 31.36 percent.<sup>79</sup>

#### 2. The 50/50 Profit-Split Method

If, for any product, the possessions corporation elects the profit-split method, it must also have manufactured that product in the possessions. In the case of Puerto Rico, this requirement is met if: (1) the product has been substan-

tially transformed by the possessions corporation in Puerto Rico; (2) the operations conducted by the possessions corporation in connection with the product are substantial in nature and generally are considered to constitute manufacture or production; or (3) the conversion costs incurred by the possessions corporation in Puerto Rico (including direct labor, factory burden, and testing of components) account for 20 percent or more of the total cost of goods sold by the possessions corporation. In this context, packaging, labeling, and minor assembly operations are not deemed to constitute the manufacture or production of product.80

Under the profit-split option, 50 percent of the combined taxable income of the possessions corporation and its U.S. affiliates, as derived from "covered sales" of the possession product, is allocated to the possessions corporation.<sup>81</sup> The remainder of the combined taxable income is generally

<sup>76</sup>IRC section 936(h) distinguishes be-

tween these forms of intangible property.

"Manufacturing intangibles" refers to any

#### Example No. 3

The facts are the same as Example No. 2 except XYZ manufactures product D under a license from an unrelated person. XYZ pays the unrelated party an annual license fee of \$25,000. Consequently, the worldwide product-area research amount increases to \$525,000.

patent, invention, formula, process, design, or know-how. "Marketing intangibles" includes any intangible property defined in IRC section 936(h)(3)(B), if it is used in marketing a product. Therefore, a determination must be made under the cost-sharing option as to what portion of the final sales price of the possession product constitutes a return to manufacturing intangibles (and is therefore tax-exempt income to the possessions corporation) and what portion is a return to marketing intangibles (and is therefore taxable income to the affiliates that perform the marketing). Regulations under IRC section 482

are applied to make this determination.  $^{77}$ IRC section 936(h)(5)(c)(i)(I).

 $<sup>^{78}</sup>$ Treas. reg. section 1.936-6.

<sup>&</sup>lt;sup>79</sup>In neither example may the payment be less than the payment that would be required under IRC sections 367(d) or section 482 if the electing corporation were a foreign corporation.

<sup>&</sup>lt;sup>80</sup>R.J. Boles, "Tax Incentives," p. 130.

<sup>&</sup>lt;sup>81</sup>"Covered sales" are sales by members of the affiliated group (other than foreign affiliates) to foreign affiliates or to unrelated persons. See Treas. reg. section 1.936-6.

XYZ, a possessions corporation, manufactures 200 units of possession product S. XYZ sells 100 units of S to an unrelated person in an arm's length transaction for \$10 per unit. XYZ sells the remaining 100 units to its U.S. affiliate, A, which leases the 100 units to unrelated persons. The combined taxable income for the 200 units of S is determined as follows:

#### Sales

1. Total sales by XYZ to unrelated persons (100 x \$10)	\$1,000
2. Total deemed sales by A to unrelated persons (100 x \$10)	1,000
3. Total gross receipts	\$2,000
Total Costs	
4. Total expenses <sup>1</sup>	\$1,200
Combined Taxable Income and Allocation of Income Attributable to the 200 Units of S	
5. Combined taxable income (line 3 minus line 4)	\$800
6. Share of combined taxable income apportioned to XYZ (50% of line 5)	400
7. Share of combined taxable income apportioned to A	
(line 5 minus line 6)	400

<sup>&</sup>lt;sup>1</sup>Research, development, and experimental costs are the higher of (1) the research and development allocation under section 861 or (2) 120 percent of total research costs multiplied by the ratio of sales by the possessions corporation to total sales.

allocated to U.S. affiliates. For purposes of computing the combined taxable income from the possession product, all direct and indirect expenses relating to the product are taken into account, including income attributable to both manufacturing and marketing intangibles associated with the product. The combined taxable income is computed separately for each product produced, or type of service rendered, by the possessions corporation in the possession.

Example No. 4 shows how to determine the combined taxable income under the profit-split method. It is based on a Treasury Department example. <sup>82</sup> The combined taxable income in Example No. 4 (\$800) is the total gross receipts from the possession product (\$2,000) minus the total expenses attributable to the development and production of this product (\$1,200). The income from the subsequent leasing of the 100 units by A to unrelated persons is attributed entirely to A.

#### IV. The Economic Impact of Section 936 on Puerto Rico

#### A. The Broad Economic Trends

The special tax credit afforded to U.S. corporations operating in the possessions clearly was helpful in promoting Puerto Rican economic growth in the 1950s and the 1960s. During this period, the credit was instrumental in transforming Puerto Rico from an agricultural economy to one primarily based on manufacturing. Puerto Rico became the "economic miracle" of the Caribbean, as real GNP per capita rose at an average annual rate of over 5 percent, compared to an annual rate of 2.2 percent for the United States during the same period.83

Since the mid-1970s, however, the section 936 tax incentives have proved to be both ineffective and inefficient as a vehicle for sustained economic growth. Consistent with this conclusion are three telling concerns. First, both em-

ployment and new physical investment in Puerto Rico have stagnated. Second, the composition of section 936 production has declined in labor intensity and has become increasingly capitalintensive. Third, the total cost of the tax credit to the U.S. Treasury has increased substantially. Collectively, these trends indicate that the limited benefit of the incentive to the Puerto Rican employee is increasingly unjustifiable in relation to the tax revenues foregone by the deficitplagued U.S. Treasury.

Manufacturing employment in Puerto Rico virtually stagnated during 1970-86, and total nongovernment employment remained steady or declined throughout 1974-83.84 The island is currently experiencing very high unemployment (18.1 percent), low labor-force participation (45.7 percent), and a high rate of migration to the mainland in search of jobs.85 Similarly, during the past two decades, aggregate physical investment in Puerto Rico has remained stagnant. The annual rate of investment declined sharply during the 1970s and early 1980s, and total fixed annual investment is only

<sup>82</sup>Treas. reg. section 1.936-6(1)(b), Q&A 11. On January 11, 1994, the IRS proposed controversial regulations under section 936(h) that amend the rules under the profit-split method for determining combined taxable income attributable to a possessions product that is a component product or an end-product form. For a summary of the proposed regulations (IL-068-92), see 8 Tax Notes Int'l 226 (January 24, 1994). For coverage of an IRS public hearing on the proposed regulations, see 9 Tax Notes Int'l 129 (July 18, 1994).

<sup>&</sup>lt;sup>83</sup>U.S. Department of the Treasury, "Sixth Report," p. 19.

<sup>3411</sup> 

<sup>&</sup>lt;sup>85</sup>National Bureau of Labor Statistics, by phone, August 1993. This rate of migration is currently hovering around 1 percent per year. See J.T. Hexner et al., Puerto Rican Statehood, p. 5.

now approaching the levels of the early 1970s.86

The change in the composition of section 936 corporations parallels this trend and is equally dramatic. Specifically, the share of section 936 activity during the past three decades in such laborintensive industries as textiles has diminished significantly, while the share in capital- and technology-intensive industries such as pharmaceuticals has increased commensurately. In 1960, for example, chemicals and machinery, two very technology-intensive industries, made up 22 percent of the net manufacturing income in Puerto Rico: by 1989. that share had increased to over 73 percent.87 Clearly, capital-intensive firms—rather than the labor-intensive industries that section 936 was designed to attract—have made the most use of the provision.

The part of the section 936 tax incentive that goes toward wages could be the most meaningful contribution of external capital to the economic health of Puerto Rico. With the high level of capital intensity, the ratio of wages and salaries to the total value added of section 936 firms is low. One indicator of this is the ratio of proprietors' income (profits, interest, etc.) to total value added for section 936 firms. In 1991, this figure was 92.3 percent for the pharmaceutical industry and 81.2 percent for the electrical machinery industry. These two industries collectively account for over 60 percent of the entire section 936 credit. Therefore, for those corporations that benefit from the majority of the incentive, wages, and salaries accounted for less than 20 percent of the total value added.

In light of stagnant employment and investment on the island and the declining labor intensity of section 936 industries, the concern, therefore, is the increasing cost-ineffectiveness of section 936. In 1989, the average revenue

cost of the tax credit per employee in a section 936 corporation was \$22,375. Before-tax annual wages for the year were, however, only \$20,540. Hence, the federal government paid approximately \$1.08 for each dollar paid to employees of section 936 corporations. Section 936 is also ineffective with respect to its low impact on physical investment, as measured by total assets of section 936 corporations per dollar of foregone tax revenues. In 1989. the total assets of section 936 manufacturing firms amounted to \$5.9 billion. Given the \$2.5 billion tax revenue cost of the program in that year, it would take less than 2.5 years for the value of foregone tax resources to equal net assets. Put simply, raw cost-effectiveness would have supported buying the section 936 manufacturing plant and equipment and literally giving it away to the corporations to operate, rather than prolonging the tax credit.

The problems with section 936 are most evident in the pharmaceutical industry. For the period 1980-90, the amount of estimated income exempt from taxes for 26 pharmaceutical firms examined by the General Accounting Office (GAO) totaled about \$24.7 billion.88 This translates into an estimated total tax savings of about \$10.1 billion in 1990 dollars for the Puerto Rican operations of these firms. 89 In 1989, however, the total assets of the pharmaceutical industry in Puerto Rico were only \$2.53 billion. Perhaps the strongest indicator of the profitability of the Puerto Rican operations of these pharmaceutical firms is that 17 of the 21 most-prescribed drugs in the United States were authorized for manufacture in Puerto Rico.<sup>90</sup> One senator has stated that the GAO document "undeniably demonstrates that the American government has given the pharmaceutical industry a blank check to pillage the federal Treasury through the section 936 tax credit."91

#### B. The Relationship Between Section 936 and Transfer Pricing Abuse: The Results of the TEFRA Amendments

The cost-ineffectiveness of section 936 generally testifies to the ineffectiveness of the TEFRA amendments. These amendments were supposed to limit substantially the amount of profits a possessions corporation could claim as tax-free earnings from the transfer of intangible assets. The facts show that this goal has not been met.

Indeed, the data since 1982. the year in which the amendments were promulgated, show the continued role of transfer pricing in artificially increasing the profit rates of the possessions corporations. For example, in 1983. the reported before-tax annual rate of return on operating assets for manufacturing corporations participating in the section 936 program was 54.1 percent, more than five times the rate of return for mainland manufacturing operations (10.3 percent).92 If the true rate of return for section 936 investments in Puerto Rico were this high, firms would have strong incentives to increase their real investment on the island, and investment would be booming.

<sup>&</sup>lt;sup>86</sup>Id., pp. 25-26.

<sup>87</sup>U.S. Congressional Budget Office, "Potential Economic Impacts of Changes in Puerto Rico's Status under S.712," report prepared for the U.S. Senate Committee on Finance, April 1990, Table 3.

<sup>&</sup>lt;sup>88</sup>U.S. General Accounting Office, "Pharmaceutical Industry," p. 21.

<sup>&</sup>lt;sup>89</sup>Id.

<sup>&</sup>lt;sup>90</sup>Id., p. 6.

<sup>&</sup>lt;sup>91</sup>Richardson, "Pryor Blasts Drug Company Tax Breaks With GAO Ammunition," 4 Tax Notes Int'l 1093 (May 25, 1992).

<sup>&</sup>lt;sup>92</sup>U.S. Department of the Treasury, "Sixth Report," p. 36.

This has not been the case, however. 93 In 1988, for example, total fixed investment in Puerto Rico was about 20 percent of GNP. compared to 25 percent in the 1966-73 period. What this suggests is that the TEFRA amendments are not blocking the transfer by corporations of large amounts of income into Puerto Rico, the Puerto Rican source generation of which is attributable to factors that are unrelated to real investments in Puerto Rico. Yet, the section 936 lobby was able to convince the Reagan administration to continue to rely on the section 936 tax subsidy as a development tool for the Caribbean Basin.

#### V. Section 936 and the Caribbean Basin Initiative

#### A. Targeting Section 936 at the Broader Goals of Regional Development

The Caribbean Basin Initiative (CBI) was introduced by the Reagan administration in 1983 to allow qualified Caribbean countries to trade on more favorable terms with the United States.94 This should have worked to increase exports to the United States from CBI countries. The Tax Reform Act of 1986 served to integrate section 936 with this development initiative. Prior to the 1986 act, section 936 allowed the active income earned by a possessions corporation to be invested tax-free in certain eligible activities in Puerto Rico and other U.S. possessions.95 The income earned from these investments is referred to as "qualified possession source investment income" or QPSII. The 1986 act expanded the area in which investments could be made to include the U.S. Virgin Islands and qualified CBI countries, as long as the investments were made through qualified financial institutions.96 The income generated by such investment qualifies as QPSII and is exempt from U.S. tax. A similar exemption from Puerto Rican tax applies under Puerto Rican law.97

The 1986 act imposes a number of requirements regarding when earnings of section 936 firms will qualify for investment in a CBI country or possession. The first requirement is that investments can only be made in qualified Caribbean Basin countries as designated under the Caribbean Economic Recovery Act of 1983.98 Twenty-three countries have thus far qualified and are so designated.99 To be eligible for these tax-exempt investments, CBIqualified countries are required to enter into a Tax Information Exchange Agreement (TIEA) with the United States.100 The purpose of the TIEA is to allow the United States and CBI governments to share tax and other information that could lead to the arrest of drug traffickers, tax evaders, and other criminals.101

Another requirement for these investments is that they be in "active business assets" or "development projects." The Senate Finance Committee report that accompanied the CBI amendment to section 936 defines these as follows:

A development project generally means an infrastructure investment, such as a road or water treatment facility, that directly supports industrial development. Active business assets generally means plant, equipment, and inventory associated with a manufacturing operation. <sup>103</sup>

Treasury Department regulations further define these terms so that qualified investment is generally permitted in tangible property used in a trade or business in qualified CBI countries, including reasonable incidental expenditures (such as installation costs).<sup>104</sup>

A section 936 company cannot receive a tax exemption if it invests funds directly in an otherwise-qualified CBI project. Instead, the section 936 company must invest through a "qualified financial institution." The Government Development Bank for

Puerto Rico and the Puerto Rico Economic Development Bank are both defined as qualified financial institutions. Other than those two, a financial institution in Puerto Rico may qualify if it is: (1) a "banking, financing, or similar business" that is "organized under the laws of the Commonwealth of Puerto Rico or is the Puerto Rican branch" of such a business and is an eligible depository institution for investments from section 936

<sup>93</sup> Hexner et al., *Puerto Rican*Statehood, for a discussion of why section
936 is incompatible with Puerto Rico's sustainable economic development and why
statehood presents a much more efficient
vehicle for continued growth.

<sup>&</sup>lt;sup>94</sup>The Caribbean Basin Initiative is the common name of the Caribbean Basin Economic Recovery Act, Pub. L. No. 98-67, 97 Stat. 384 (1983) (codified as amended in scattered sections of 19 U.S.C. and 26 U.S.C.). Under the act, qualified countries receive a reduction or elimination of tariffs on certain products, along with access to relatively low interest rate loans, provided with certain section 936 funds.

<sup>95</sup>IRC section 936(d)(2).

 $<sup>^{96}</sup>IRC$  section 936(d)(2)(B) and 936(d)(4).

<sup>&</sup>lt;sup>97</sup>13 L.P.R.A. section 256a(2)(j)(A).

<sup>98</sup>IRC section 936(d)(4).

<sup>99</sup>Those countries and possessions are Antigua and Barbuda, Aruba, the Bahamas, Barbados, Belize, British Virgin Islands, Costa Rica, Dominica, Dominican Republic, El Salvador, Grenada, Guatemala, Guyana, Haiti, Honduras, Jamaica, Montserrat, Netherlands Antilles, Panama, St. Kitts and Nevis, St. Lucia, St. Vincent and the Grenadines, and Trinidad and Tobago. Nicaragua has requested designation, and the U.S. is currently reviewing that request.

 $<sup>^{100}</sup>$ IRC section 936(d)(4)(B).

<sup>101</sup>Flax-Davidson, "Tax-Exempt Investment for the Caribbean Basin Initiative Region," International Lawyer 25, no. 4 (Winter 1991): 1025.

<sup>&</sup>lt;sup>102</sup>IRC section 936(d)(4)(A)(i).

 <sup>103</sup>U.S. Senate, Committee on Finance,
 Tax Reform Act of 1986, S. Rep. No. 313,
 99th Cong., 2d Sess. 384 (1986).

<sup>104.</sup> Requirements for Investments to Qualify under Section 936(d)(4) as Investments in Qualified Caribbean Basin Countries," Treas. Reg. section 1.936-10(c)(4), (5) (May 10, 1991).

<sup>105</sup>IRC section 936(d)(4)(A).

firms, as qualified by the commissioner of financial institutions under Puerto Rican regulations; (2) "such other entity as may be determined by the commissioner"; or (3) a "single-purpose entity" established in Puerto Rico as an eligible institution solely to invest funds from section 936 firms in qualified CBI assets. 106

All lending of these funds to a qualified CBI recipient must be approved by the commissioner of financial institutions for Puerto Rico. 107 Additionally, the recipient of CBI funds must open its books to the United States and Puerto Rican governments to assure that the funds are being used in accordance with the law. 108 A 1990 congressional amendment to section 936 requires the government of Puerto Rico to ensure that at least \$100 million is invested annually in qualified CBI investments. 109

#### B. Evaluation of the CBI: A Weak Justification for Prolonging Section 936

Even among those who acknowledge the transfer pricing abuses of section 936 firms, there are proponents who justify the continued extension of section 936 benefits because of the tax credit's role in the CBI program. They argue that the elimination or any reduction in the section 936 tax credit would proportionately damage the CBI because of the close integration of the two programs. It has been asserted, for example, that at least \$500 million of qualified funds have been appropriately invested under this program, creating close to 20,000 jobs in the CBI countries and more than 2,500 jobs in Puerto Rico. 110 These statistics have been used to support the claim that the program is functioning as intended and that section 936 should remain untouched.

Other evidence, however, suggests that the CBI program has been unsuccessful. The Latin American and Caribbean Economic Commission reported an

average 17.2-percent reduction in per capita gross domestic product during the 1980s.111 Latin America and the Caribbean also experienced a 0.8-percent decrease in real gross national product in 1990, and record loans in that vear added to their already staggering foreign debts.112 Thus, in relative terms, the very nations the CBI was intended to support have been steadily losing ground. 113 The claim of positive long-run development impact from the \$500 million of CBI funds purported to have been allocated and the 20,000 jobs created in the CBI countries is dubious at best.

Despite the preferential treatment offered to CBI countries under the program, there has been a constant decline in the value of U.S. imports from these countries. U.S. imports from CBI countries reached an all-time high in 1983, the year in which the program was enacted. 114 Between 1983 and 1986, however, exports from the CBI countries to the United States declined by a total of 31 percent. 115 By contrast, the level of U.S. exports to these countries has remained steady.116 According to the U.S. International Trade Commission:

In 1986, for the first time in a number of years, the United States had a small surplus with the Caribbean countries collectively, making the basin one of the few areas in the world with which no U.S. trade deficit was recorded. This was the result of a significant decline in U.S. imports from the Caribbean Basin, from \$9.0 billion in 1983 to \$6.2 billion in 1986, while U.S. exports to the area remained approximately the same, fluctuating around \$6.0 billion.117

Also, in 1986, U.S. imports from CBI countries represented only 1.7 percent of the total U.S. imports, while U.S. exports to these countries represented 3 percent of the total U.S. export market. 118

Recent data suggest a continuation of these trends. In 1990, U.S. imports from CBI countries were approximately \$1.4 billion less than in 1983. <sup>119</sup> This constitutes a 2.1-percent annual rate of reduction in the amount of imports and a 16-percent gross decline. <sup>120</sup>

Indeed, the CBI might be judged as a program of phantom benefits and phantom results. Over 93 percent of the exports generated from Caribbean countries designated under this program already entered the United States duty-free prior to the enact-

 $<sup>^{106}</sup>$ Treas. Reg. section 1.936-10(c)(3).

<sup>&</sup>lt;sup>107</sup>IRC section 936(d)(4)(A)(ii).

<sup>&</sup>lt;sup>108</sup>IRC section 936(d)(4)(C)(iii).

<sup>&</sup>lt;sup>109</sup>See IRC section 936(d)(4)(D) (West Supp. 1991) (effective for calendar years after 1989); H.R. 1594, 101st Cong., 2d Sess., 136 Cong. Reg. H5887, H5896 (daily ed. July 30, 1990).

<sup>&</sup>lt;sup>110</sup>Price Waterhouse, "Section 936 Report, Volume 1: Benefits and Costs of Section 936," prepared for Puerto Rico, U.S.A. Foundation, May 1991, Table IV.B. See also R.J. Sierra Jr., "Funding Caribbean Basin Initiative Activities with Section 936 Funds," International Tax Journal (Spring 1992): 57-58.

<sup>111.</sup> Mexico, Central American Countries Plan Free Trade Agreement to Be Reached by 1996," *Intl. Trade Rep.* (BNA) 8, no. 3 (Jan. 16, 1991): 87.

<sup>112</sup>c<sub>4</sub>Latin American Economies Register Decline of 0.8 Percent in 1990, IDB Report Shows," *Intl. Trade Rep.* (BNA) 8, no. 15 (Apr. 10, 1991): 554.

<sup>&</sup>lt;sup>113</sup>J.C. Malloy, "The Caribbean Basin Initiative: A Proposal To Attract Corporate Investment and Technological Infusion via an Inter-American Protection for Intellectual Property," University of Miami Inter-American Law Review 23, No. 1 (1991): 184.

<sup>&</sup>lt;sup>114</sup>U.S. International Trade Commission, "Annual Report on the Impact of the Economic Recovery Act on U.S. Industries and Consumers, Second Report 1986," September 1987, p. 6.

<sup>&</sup>lt;sup>115</sup>Id., p. 8.

<sup>&</sup>lt;sup>116</sup>Id., p. 1.

<sup>117</sup>Id.

<sup>118&</sup>lt;sub>Id</sub>

<sup>&</sup>lt;sup>119</sup>U.S. Department of Commerce, Guidebook: Caribbean Basin Initiative (1991), p. 55.

<sup>120</sup>Id.

ment of the CBI. 121 In addition, the elimination of already low U.S. tariffs (generally ranging from 5 to 7 percent) on Caribbean industrial products does not make these products significantly more competitive in the U.S. market. 122 Moreover, the CBI excludes from its list of qualified products a number of items produced by the most labor-intensive industries, including apparel and leather goods.

What the CBI program has successfully done, however, is expand the scope of political leverage for section 936 companies by broadening the scope of their potential investment. As dollars have been invested in more CBI countries, section 936 companies have gained increasing clout with these countries and enlisted their help in lobbying against the curtailment of the tax credit. Nevertheless, because of the lack of real benefits from the CBI and because the actual amount of imports from the CBI countries has been steadily decreasing, while U.S. exports have remained steady, the continued existence of the section 936 tax credit cannot be justified by linking it to the CBI program.

#### VI. Further Reforms Affecting Section 936

#### A. Effects of Section 482 Regulations on Section 936

Section 482 of the Treasury regulations provides most of the guidelines concerning the proper allocation of income in a transfer pricing transaction. On July 1, 1994, final regulations were issued under section 482 that contain provisions that alter the manner in which transfer prices for intangible property will be reviewed by the IRS and that specifically coordinate with the transfer pricing requirements of section 936. The final section 482 regulations provide for greater taxpayer flexibility, at the cost of more stringent documentation requirements.123 The IRS anticipates that this will diminish the number of disputes between the IRS and taxpayers. Some practitioners contend, however, that the policy, in its move toward greater flexibility, imposes an unmanageable administrative burden on the IRS.

The regulations reaffirm the applicability of the arm's length standard and continue to emphasize analysis that relies on the structure and circumstances of the individual transaction. However, added flexibility now comes via the applicability of a range of acceptable arm's length results as opposed to a single arm's length price. Also, consistent with the reality of varying market conditions and transaction circumstances, there is now no strict priority of pricing methods. Instead, the accuracy of the pricing method, with respect to the case in question, decides its appropriateness. In another move towards taxpayer flexibility, the prices actually charged in controlled transactions need not reflect the arm's length price that must be reported on the income tax return. Where reported price differs from the price actually charged, compensating adjustments are made to reflect the disparity.124 Finally, the standards that must be met before transactions are considered comparable have been relaxed. Under all pricing methods, a reasonable number of adjustments is permitted where transactions are not exactly comparable.

With regard to possessions corporations, section 482 regulations provide that when a controlled taxpayer has elected for cost-sharing under section 936(h), the amount of the cost-sharing payment that is required under this section will not be less than the payment that would be required under the section 482 regulations (if the electing taxpayer were a foreign corporation). Also, the 936 corporation must apply the section 482 pricing methods for intangibles before giving effect to the

provisions that treat the 936 corporation as the owner of this property.

One reviewer of the tax changes makes the following claim:

It is almost impossible using the arm's length method of section 482 of the Internal Revenue Code to determine accurately the tax obligations of multinational corporations dealing only in tangible goods: it is impossible to do so when these corporations are earning money from intangibles. . . . In short, the IRS's section 482 enforcement efforts are unworkable because the system is too complex, cumbersome and expensive to catch all but the most blatant tax evaders. 125

This claim is troubling given the findings of a 1993 Ernst & Young study, which estimated that the government's transfer pricing initiatives would collect less than 10 percent of Treasury Department projections. 126

In effect, the complexity of the section 482 regulations has the potential to render them unadministerable. Indeed, the true price of taxpayer flexibility is that the circumstances of transfer pricing arrangements will gain in subjectivity and will increasingly call for judgment on a case-by-case basis. Accordingly, cases involving highly differentiated products, for which benchmark arm's length transactions are not

<sup>&</sup>lt;sup>121</sup>T.L. Raleigh, "The U.S. Caribbean Basin Initiative," *International Business Lawyer* 15, no. 3 (March 1987): 137.

<sup>1227</sup> 

<sup>&</sup>lt;sup>123</sup>J. Turro, "U.S. Releases Final Transfer Pricing Regulations Under Section 482," 9 Tax Notes Int'l 79 (July 11, 1994).

<sup>&</sup>lt;sup>124</sup>Reg. section 1.482-1(e)(2).

<sup>125</sup> Lobel, Banta, and Gueron, "Barclays: A Test of the Administration's Willingness To Collect Taxes from Multinational Corporations," Tax Notes, June 28, 1993, p.1841.

 $<sup>^{126}</sup>Id.$ 

XYZ is a possessions corporation operating in the 1998 tax year with an active business income from possession-based operations of \$900,000. QPSII is \$100,000. With no section 936 tax credit, U.S. tax liability on this income would amount to \$315,000 and \$35,000, respectively. The corporation's section 936 credit would be limited to \$161,000 (40 percent of \$315,000 plus a full credit on the QPSII tax liability).

Further, XYZ incurred \$60,000 in possession taxes. A partial deduction of possession income tax is permitted. This is calculated as the total in possession income tax (\$60,000) multiplied by the ratio of (a) total U.S. income tax liability with no section 936 credit less the amount of the credit (\$350,000 - \$161,000 = \$189,000) to (b) the total U.S. income tax liability with no section 936 credit (\$350,000). The deduction in this case would be \$32,400 (\$60,000 x \$189,000/\$350,000). This reduces total taxable income to \$967,600. Hence, the pre-section-936-credit tax liability is \$338,660, and the post-credit liability is \$177,660 (\$338,660 - \$161,000).

Start of Tax Year	Percentage Limitation
1994	60
1995	55
1996	50
1997	45
1998, and thereafter	40

A taxpayer that utilizes the percentage limitation is permitted a deduction for a portion of its possession income taxes paid or accrued during the tax year. The

easily identified (more often true of intangibles), will rely on extensive cost, pricing, and market data—often from unwilling competitors.

In practical terms, then, the auditing requirements of the policy leave the short-staffed IRS at a disadvantage compared to the multinational corporations with their batteries of highly paid lawyers, accountants, and economists.

#### B. Further Limitations on the Section 936 Tax Credit

As a result of concerns about transfer pricing abuse by pharmaceutical and other capital-intensive firms and because of the low levels of employment-producing investments made by section 936 firms, section 936 has been increasingly opposed by the U.S. Treasury Department and members of Congress. Indeed, on December 31, 1993, legislation designed to curb transfer pricing abuses and increase levels of investment in employment-producing activities, enacted as part of the Omnibus Budget Reconciliation Act of 1993, came into effect. 127

Under the new legislation, the section 936 credit will be calcu-

lated in a manner consistent to that used prior to December 31, 1993.128 However, the amount of the credit will then be limited in one of two ways, 129 with the choice of method left to the taxpayer. The first, the percentage limitation, limits the credit by a statutorily defined percentage (that decreases in future years) of the section 936 credit allowable under present law. The second alternative, the economic activity limitation, links the limitation on the credit to a composite of factors that serve as proxy for the firm's level of economic activity in the possessions. All affiliated130 possessions corporations are required to choose the same creditlimitation alternative. 131

## 1. The Percentage Limitation

Under the percentage limitation, the section 936 credit allowed to a possessions corporation against U.S. tax on business income for a tax year is limited to a specific percentage of the credit that would be permitted under the laws prior to the 1993 revision. A five-year transition rule governs the phase-in. The percentages are: 132

129In a measure to support Puerto Rican tax revenues, given the credit limitations, the revised legislation temporarily increases the cover-over of rum excise taxes to Puerto Rico and the Virgin Islands from \$10.50 to \$11.30 per proof gallon. The increased cover-over rate applies through 1998.

130The consolidated return rules are used to determine whether a possessions corporation is part of an affiliated group. However, stock owned by attribution under the rules of IRC section 1563 is treated as if it were owned directly, and the exclusions from the definition of "includible corporation" listed in IRC section 1504(b) are disregarded.

131 Should a possessions corporation that employs the percentage limitation become a member of a group that uses the economic activity limitation, then the first corporation will be deemed to have revoked its election to use the percentage limitation. The Treasury secretary is authorized to develop regulations to treat two or more possessions corporations as members of the same affiliated group in order to prevent avoidance of the consistency rule.

 $<sup>^{127}</sup>See$  H.R. 2264, 103rd Congress, 1st Sess

<sup>128</sup> Under the new legislation, there is a new separate foreign tax credit limitation category for computing the alternative minimum tax (AMT) foreign tax credit. The new category includes the portion of dividends received from a possessions corporation for which the dividends-received deduction is generally disallowed, and thus is included in alternative minimum taxable income.

<sup>132</sup>IRC section 936(a)(4)(B)(ii).

XYZ is a possessions corporation that elects to use the economic-activity limitation. XYZ does not choose the profit-split method for computing its income from intangibles. Wage and fringe benefit expenses for XYZ total \$180,000 (\$150,000 in qualified possession wages and \$30,000 in employee health, accident, and life insurance plans). XYZ's depreciation deductions amount to \$50,000 for short-life tangible property, \$30,000 for medium-life tangible property, and \$20,000 for long-life tangibles. XYZ has \$1,000,000 of taxable income for the year. Nine hundred thousand of this is income from active business operations. Of the remaining \$100,000, \$50,000 is QPSII and \$50,000 is other taxable income. Sixty thousand dollars is paid in possession income taxes.

Under the laws in effect through the end of 1993 (assuming no deduction for possession income taxes), the section 936 credit amounts to \$332,500 (35% of \$950,000 in total income less other taxable income). U.S. tax liability equals \$17,500 (35% of \$50,000 in other taxable income).

The revised section 936 law does not change the credit attributable to QPSII. Thus, \$17,500 (35% of \$50,000 in QPSII) of the present law credit is not subject to the economic activity limitation. The remainder, \$315,000 is, however, subject to the limitation.

#### Qualified Compensation

Qualified possession wages amount to \$150,000. Potentially, \$25,000 in fringe benefit expenses ( $$150,000/$180,000 \times $30,000$ ) could have been included in the credit limitation base. The 15% limitation on fringe benefits applies, however, limiting the allocable amount to \$22,500 (15% of \$150,000). Total qualified compensation thus amounts to \$172,500 (\$150,000 + \$22,500), 60 percent of which is \$103,500.

#### Depreciation Deductions

The depreciation component of the credit limitation is the sum of (1) 15% of the \$50,000 depreciation allowance on short-life property, (2) 40% of the \$30,000 depreciation allowance on medium-life property, and (3) 65% of the \$20,000 depreciation allowance on long-life property, for a total of \$32,500.

#### Possession Income Taxes

None of the \$60,000 of possession income taxes (a 6% effective rate) is disqualified from the credit limitation base by virtue of the maximum 9% effective tax rate provision. However, only the portion of the \$60,000 that is allocated to nonsheltered income may be included in the credit limitation base. This portion is a function of the ratio of the increase in tax as a result of the compensation and depreciation limitations, and the tax that would be paid in the absence of the section 936 tax credit.

In the absence of the compensation and depreciation limitations, XYZ's U.S. tax liability would be \$17,500. With the limitations, it would amount to \$350,000 (35% of \$1,000,000) less (1) \$136,000 (\$103,500 + \$32,500), the active business section of the 936 credit and (2) the QPSII credit of \$17,500. That is, \$196,500. Hence, the increase in tax liability is \$179,000 (\$196,500 - \$17,500).

With no section 936 credit, the U.S. income tax liability would amount to \$350,000 (35% of \$1,000,000).

The amount of possession income taxes which may be included in the credit limitation base is therefore \$30,686 [(\$179,000/\$350,000) x \$60,000].

#### Total Economic Activity Limitation

The total limitation on the active business credit is therefore \$166,686 (that is, \$103,500 for compensation, plus \$32,500 for depreciation, plus \$30,686 for possession income taxes) compared to \$315,000 under the regulations before revision. The full credit of \$17,500 on QPSII is also granted. The corporation's net U.S. tax liability is therefore \$165,814 (\$350,000 - \$166,688 - \$17,500).

 $^1$ Taxable income is computed in accordance with the pre-December 31, 1993 rules for determining the taxable income of a possessions corporation.

portion of the taxes so deductible is the portion that is allocable to the corporation's taxable income, the U.S. tax on which is not offset by the section 936 credit as a result of the limitation.

The operation of the percentage limitation is shown in Example No. 5. As the example shows, the limitation clearly reduces the section 936 credit over time. However, a firm's choice to use this alternative will be a function of the magnitude of its potential credit in relation to the credit available under the economic activity limitation. This, of course, will be determined by the firm's capacity to claim credit from the expansion of new laborintensive activities, as well as activities that purport to be so.

### 2. The Economic Activity Limitation

The sum of three proxy measures for economic activity in the possessions serves as an upper limit for the tax credit allowed to a possessions corporation for a tax year. The credit against U.S. tax on the possessions corporation's business income may not exceed the sum of the following components: 133

- 60 percent of qualified compensation;
- the applicable percentages of depreciation deductions on qualified tangible property claimed for regular tax purposes by the corporation; and
- a portion of the possession income taxes incurred during a given year, if the corporation does not elect the profit-split method to allocate income from intangibles.

U.S. tax liability, therefore, is computed by subtracting the sum of the above three components from the amount of precredit U.S.

<sup>&</sup>lt;sup>133</sup>IRC section 936(a)(4)(A).

tax that, under general circumstances, would be owed.

#### a. Qualified Compensation

The first component of the economic activity limitation is 60 percent of qualified compensation. Qualified compensation is the sum of:134 (1) the aggregate amount of the possessions corporation's qualified possessions wages for the tax year 135 and (2) allocable employee fringe benefit expenses for the tax year. 136 Qualified possessions wages are defined as wages paid or incurred by the possessions corporation during the tax year to any employee for services performed in a possession. 137 However, such services must be performed while the principal place of employment of the employee is within that possession.

#### b. Depreciation Deductions

The second component is the sum of the following applicable percentages of allowable depreciation deductions:<sup>138</sup>

- (1) 15 percent of the depreciation deductions allowable to shortlife qualified tangible property;
- (2) 40 percent of the depreciation deductions allowable to medium-life qualified tangible property; and
- (3) 65 percent of the depreciation deductions allowable to longlife qualified tangible property.<sup>139</sup>

#### c. Possession Income Taxes

The final component of the economic activity limitation is a portion of the income taxes paid or incurred to a possession by corporations that do not elect the profit-split method. <sup>140</sup> Possession income taxes paid in excess of a 9-percent effective rate of tax are not included. <sup>141</sup> Moreover, only the portion of taxes that satisfies this effective rate requirement and that is allocable to non-sheltered income is included. <sup>142</sup>

The operation of the economic activity limitation is shown in Example No. 6.

d. Election To Treat Affiliated Corporations as One Corporation

For purposes of computing the economic activity limitation, an affiliated group of corporations may elect to treat all affiliated possessions corporations as one corporation. For a group so electing, the available consolidated credit amount is to be allocated among the possessions corporations of the group under rules prescribed by the Treasury secretary. Any election to consolidate applies to the tax year for which such election is made and to all succeeding tax years unless revoked with the consent of the Treasury secretary.

#### e. Analysis of the Economic Activity Limitation

In concept, over the course of five years, section 936 credits will be effectively linked to growth in employment wages and tangible investment. From a practical standpoint, however, the policy is less promising.

The correlation between tax avoidance and development strategies based on complex tax incentive schemes is well established. Indeed, the compounding negative results of repeated efforts by Congress to patch the loopholes of section 936 verify this correlation.

The revisions to section 936 made by the Omnibus Budget Reconciliation Act of 1993 significantly increase administrative by the Treasury secretary. The bill does not include as qualified possession wages amounts paid to employees who are assigned by the employer to perform services for another person, unless the principal trade or business of the employer is to make employees available for temporary periods to other persons in exchange for compensation.

 $^{136}$ Fringe benefits may include: (1) employer contributions under a stock bonus. pension, profit sharing, or annuity plan; (2) employer-provided coverage under any accident or health plan for employees; and (3) the cost of life or disability insurance provided to employees. Fringe benefit expenses do not include any amount that is treated as wages. Allocable employee fringe benefit expenses are equal to a fraction of the aggregate amount that is consistent with the conditions listed above. The numerator of this fraction is the aggregate amount of the possessions corporation's qualified possessions wages (as defined above). The denominator is the aggregate amount of compensation (wages and benefits) paid or incurred during the tax year. Fringe benefit expenses may not, however, exceed 15 percent of the aggregate amount of qualified possession wages for that year.

<sup>139</sup>The terms of IRC section 168 apply to the definition and classification of depreciable tangible property.

140 Possessions corporations that utilize the profit-split method may deduct a portion of their possession income taxes paid or accrued during the tax year. This portion is the part of U.S. taxable income, the U.S. tax on which is not offset by the revised section 936 credit.

142The portion of possession income taxes allocated to nonsheltered income is determined by computing the ratio of two hypothetical U.S. tax amounts that are computed under the assumption that no credit or deduction is allowed for possession income taxes. This ratio is then multiplied by the taxable income of the corporation as computed under the assumptions that no credit or deduction is allowed for possession income taxes and that all other deductions are allowed as under present law.

The numerator of the above ratio is the U.S. tax liability of the possessions corporation that would arise under the bill by virtue of the economic activity limitation determined without any credit or deduction for possession income taxes. The denominator is the U.S. tax liability of the possessions corporation that would be imposed on the income (computed under existing section 936 rules) of the corporation without any credit or deduction for possession income taxes.

 $<sup>^{134}</sup>$ IRC section 936(a)(4)(A)(i).

<sup>135</sup>Wages for this purpose include those defined under the Federal Unemployment Tax Act (FUTA). In computing the credit limitation for a tax year, the cumulative amount of wages for each employee may not exceed 85 percent of the maximum earnings subject to tax under the Old Age Survivors and Disability Insurance (OASDI) portion of social security (currently \$57,600). Rules for making appropriate adjustments to this limit for part-time employees and employees whose principal place of employment is not within a possession for the entire tax year are to be made

<sup>137</sup>IRC section 936(i)(1)(A).

<sup>138</sup>IRC section 936(a)(4)(A)(ii).

<sup>141</sup> IRC section 936(i)(3)(A)(ii).

complexity and auditing. This, in turn, enhances the potential for tax avoidance by presenting new opportunities for tax manipulation of corporation expenses and transfers to maximize tax benefits under section 936. Payroll padding, for example, is certain to replace transfer pricing as a means of increasing the credit to 936 firms in the absence of substantive real growth in employment and investment.

In addition, since December 31, 1993, Puerto Rican firms, which pay a 42-percent income tax rate, are being forced into unfair competition with firms from the mainland. Mainland firms will benefit not only from Puerto Rican tax incentives but also from the tax credits that subsidize 60 percent of wages in section 936 firms and significant percentages of depreciation on tangible investment. Payroll and employment expansion under the revision will not be market-based and will be unsustainable in the absence of the tax credit. Accordingly, the tax credit will foster a dependence not just on the part of U.S. multinationals, but also by Puerto Rican workers. whose livelihoods will increasingly be directly dependent on revenues foregone by the U.S. Treasury.

#### VII. Conclusion

History should have taught us the following lessons:

Section 936. If a tax credit is offered based on the amount of possessions-source income a corporation generates, then methods will be found to transfer income streams from the mainland to the possession. The income streams most easily transferred will be those related to intangible assets. These assets represent little or no real investment to the possession.

Section 482 and the TEFRA Amendments. Both the TEFRA amendments and the IRS's track record in enforcing section 482 indicate that almost nothing can be done to stop intangible income transfers in either a useful development time frame (482 cases take more than 10 years to resolve), or in any but the most egregious of violations.

The Caribbean Basin Initiative. Reinvestment of section 936 profits flows readily into profitmaximizing and risk-minimizing, rather than development-maximizing, uses.

Under the current state of the law, if these lessons have been learned, the future holds the following for the possessions:

- (1) If section 936 remains, nothing can or will be done to stop the diversion of income derived from intangibles to possessions corporations.
- (2) Unless a possessions corporation determines that it will be unlikely to secure new intangible assets in the future (a proposition very unlikely in the pharmaceutical industry), the new percentage limitations on the section 936 credit will not be elected.
- (3) Possessions corporations will have plans drawn up targeting pre-existing possessionsbased labor-intensive businesses for mergers. Premium targets will have low risks but high balancesheet (wage and tangible property) attributes. These targets will not necessarily be those best suited to the long-term economic development of the possession. Similar to the CBI, these plans will be investment plans to "buy and hold," not development plans to "buy and further develop" local industries.
- (4) Very quickly after income starts to flow to a possessions intangible, the possessions corporation will implement its acquisition strategy. The possessions corporation will aggressively strive to maximize the elements of the three-factor economic activity limitation formula.
- (5) Development officials in the possessions should see ownership changes in the assets base in two steps. Initially, properties that

the possessions corporations had leased will be purchased, and delivery, cleaning, or securitytype subcontractors will be absorbed as in-house departments. Secondly, because neither the Internal Revenue Code nor regulations have any requirements linking the wages paid or the tangible property owned by the possessions corporation to the income stream that actually generates the credit, a more wide-ranging absorption of possession-based assets and wage-paying businesses will be observed.

(6) The possessions economy will stagnate. Each target absorbed will dilute the pool of possessions-based entrepreneurial talent. Each business not absorbed will struggle at a competitive disadvantage against possessions-owned competitors. Section 936 will function as its mirror opposite. In the extreme instance, section 936 will subsidize the dismantling of the Puerto Rican entrepreneurial system and the local tax base it represents.

In summary, section 936 has ceased to be an efficient means of attaining employment-producing investments in Puerto Rico and other U.S. possessions. While the initial rationale for the credit was the creation of jobs and the stimulation of economic activity in the possessions, the outcome has been far different. Firms with intangible assets now take advantage of transfer pricing laws to maximize profits without making the investments that would create sustainable growth in Puerto Rico.

The fundamental questions then are: First, can the long record of disappointment be ended? Second, can the legislation provided in the 1993 budget transform section 936 into an instrument of public benefit, rather than of private profit? We conclude that the costs of section 936 will continue to outweigh its benefits.

#### VIII. Bibliography

Aud and Fuller. "The New Temporary and Proposed Section 482 Regulations: A Wolf in Sheep's Clothing?" Tax Notes, March 15, 1993, p. 1517.

Avakian-Martin, Juliann. "TRS Acknowledges Need To Clarify Portions of Transfer-Pricing Regs." *Tax Notes*, March 23, 1992, p. 1452.

Bell and Turro. "Pryor Bill Targets Possessions Credit, Pharmaceutical Companies." *Tax Notes*, February 22, 1993, p. 1005.

Boles, Richard J. "Tax Incentives for Doing Business in Puerto Rico." *International Lawyer* 22, no. 1 (Spring 1988): 121-42.

Bradford, John J. "U.S. Possessions Corporations Returns, 1989." U.S. Department of the Treasury, Office of Tax Analysis.

Caribbean Basin Initiative: Caribbean Views; Report of a Congressional Study Mission and Symposium on the Caribbean Basin Initiative September 18-19 to the Committee on Foreign Affairs, U.S. House of Representatives. December 1987.

Caribbean Basin Initiative: Hearing before the Subcommittee on International Trade of the Committee on Finance, United States Senate, 101st Cong., 2d Sess. February 9, 1990.

Carlson, George N. "The Proposed New Transfer Pricing Rules: New Wine in Old Bottle." Tax Notes, February 10, 1992, p. 691.

Clark, Don P. "Measurement of Trade Concentration under the United States' Caribbean Basin Initiative." World Development 17, no. 6 (1989): 861-65.

Cole, Robert T. "Section 482: Proposed New Regulatory Approaches." *Tax Executive* 44, no. 2 (March-April 1992): 95-102.

Development Policy in the Caribbean: Hearings before the Subcommittees on Human Rights and International Organizations, International Economic Policy and Trade, and on Western Hemisphere Affairs of the Committee on Foreign Affairs, House of Representatives, 100th Cong., 2d Sess. July 28; August 4, 9, and 10, 1988.

Ekman, Mary Ann. "Mogle Discusses New Temporary and Proposed Section 482 Regulations at California State Bar Conference." 6 Tax Notes Int'l 243 (February 1, 1993).

Flax-Davidson. "Tax-Exempt Investment for the Caribbean Basin Initiative Region." *International Lawyer* 25, no. 4 (Winter 1991): 1021-29.

Fuller, James P. "The Proposed Section 482 Regulations." *Tax Notes Int'l* 4, no. 12 (March 23, 1992): 599-612.

Fuller, James P. "Section 482 Transfer-Pricing Regulations." Tax Notes Int'l 4, no. 9 (March 2, 1992): 438-40.

Granfield, Michael E. "An Economic and Strategic Evaluation of the Proposed Section 482 Regulations." *Tax Notes Int'l* 4, no. 24 (June 15, 1992): 1255-69.

Gray, Tim. "Two Taxwriters Push Puerto Rico To Invest More Section 936 Funds." *Tax Notes* 47, no. 13 (June 25, 1990): 1551-52.

Hanlon, Sally. "GAO Questions Commerce Department's Portrayal of Impact of CBI." *Tax Notes* 33, no. 1 (October 6, 1986): 22-23.

Hexner, J. Tomas, Glenn Jenkins, Helen F. Ladd, and K. Russell LaMotte. Puerto Rican Statehood: A Precondition to Sound Economic Growth. Cambridge, Massachusetts: Hex Inc., November 1993.

Kirchheimer, Barbara. "Clinton Tax Proposals: Some Tinkering, No Major Surprises." *Tax Notes*, April 12, 1993, p. 152.

"Latin American Economies Register Decline of 0.8 Per Cent in 1990, IDB Report Shows." *Intl. Trade Rep.* (BNA) 8, no. 15 (Apr. 10, 1991).

Lederman, Alan S. "Proposed Section 482 Regs. Adopt Comparable Profit Interval Requirement." *The Tax Magazine* 70, no. 4 (April 1992): 228-42.

Lobel, Banta, and Gueron. "Barclays: A Test of the

Administration's Willingness To Collect Taxes from Multinational Corporations." *Tax Notes*, June 28, 1993, p. 1841.

Lunger, Anilisa G. "The Caribbean Basin Initiative and the I.R.C. Section 936 Investment Program: A United States Answer to the Troubled Caribbean Region." University of Pennsylvania Journal of International Business Law 9 (1987): 741-85.

Luxner, Larry. "High Hopes for Low-Interest Loans." *Americas* 43, nos. 5-6 (1991): 22-25.

Malloy, John Cyril, III. "The Caribbean Basin Initiative: A Proposal to Attract Corporate Investment and Technological Infusion via an Inter-American System of Cooperative Protection for Intellectual Property." *University of Miami Inter-American Law Review* 23, no. 1 (Fall 1991): 175-94.

McLennan, Barbara N. "Complying With the Proposed Section 482 Regulations: How To Create and Apply a Comparable Profit Interval." 4 *Tax Notes Int'l* 1201 (June 8, 1992).

"Mexico, Central American Countries Plan Free Trade Agreement to Be Reached by 1996." Intl. Trade Rep. (BNA) 8, no. 3 (January 16, 1991).

Morrison, Philip. "Testimony before the Committee on Finance, United States Senate." April 26, 1990.

Mudd, John E. "The Impact of the Elimination of Section 936 of the Internal Revenue Code on U.S. Foreign Policy in the Caribbean." Revista de Derecho Puertorriqueno 27, no. 101 (Spring 1987): 25-45.

O'Grady, John E. "Cost-Sharing Arrangements Under the Proposed Section 482 Regulations." 4 Tax Notes Int'l 381 (February 24, 1992).

Pelzman, Joseph. "The Impact of the Caribbean Basin Economic Recovery Act on Caribbean Nations' Exports and Development." Economic Development and Cultural Change 36, no. 4 (July 1988): 753-96.

Pelzman, Joseph. "The Impact of the Caribbean Basin Economic Recovery Act on Caribbean Nations' Exports and Development: A Reply." *Economic Development and Cultural Change* 38, no. 4 (July 1990): 851-54.

Pinon, Gema M. "CBI II: Will United States Protectionist Tendencies Yield to Economic Development in the Caribbean Basin?" The University of Miami Inter-American Law Review 20, no. 3 (1989): 615-43.

Price Waterhouse. "Section 936 Report, Volume 1: Benefits and Costs of Section 936." Prepared for Puerto Rico, U.S.A. Foundation, May 1991.

Raleigh, Thomas L., III. "The U.S. Caribbean Basin Initiative." *International Business Lawyer* 15, no. 3 (March 1987): 136-38.

Richardson, Joanna. "Pryor Blasts Drug Company Tax Breaks With GAO Ammunition." 4 *Tax Notes Int'l* 1093 (May 25, 1992).

Rolfe, Robert J. "Combining the Benefits of Possessions Corporations with the Caribbean Basin Initiative." *The International Tax Journal* 16, no. 1 (Winter 1990): 5-22.

Romero, Carlos. "CBI Not Likely To Benefit From Section 936." Tax Notes 30, no. 6 (February 10, 1986): 547-49.

Santiago, Carlos Enrique. Labor in the Puerto Rican Economy: Postwar Development and Stagnation. New York: Praeger, 1992.

Sawyer, W. Charles. "The Impact of the Caribbean Basin Economic Recovery Act on Caribbean Nations' Exports and Development: A Comment on Pelzman and Schoepfle's Estimates." Economic Development and Cultural Change 38, no. 4 (July 1990): 845-49.

Sierra, Ralph J., Jr. "Funding Caribbean Basin Initiative Activities with Section 936 Funds." International Tax Journal 18, no. 2 (Spring 1992): 27-58.

Simpson, John M. "Do You See 'Convergence'? Examples of Real-World Data and the Comparable Profit Interval of Proposed Regulation Section 1.482." Tax Notes Int'l 4, no. 25 (June 22, 1992): 1311-18.

Tansill. "Puerto Rico: Independence or Statehood?" Revista del Colegio de Abogados de Puerto Rico 41 (1980): 79.

Turro, John. "Congress Asks Why Section 936 Funds Are Not Reaching the Caribbean." *Tax Notes* 47, no. 2 (April 9, 1990): 133-35.

Turro, John. "Section 482 Regulations Issued at Long Last; 'Comparable Profit' Analysis Introduced." *Tax Notes Int'l* 4, no. 5 (February 3, 1992): 207-210.

Turro, John. "Clinton Looks to Foreign Provisions for Even More Revenue." *Tax Notes* 59, no. 2 (April 12, 1993): 155-56.

Turro, John. "Possessions Credit Reform Not a Done Deal, Treasury Official Implies." *Tax Notes* 59, no. 4 (April 26, 1993): 469.

Turro, John. "Puerto Rico Officials Unveil Section 936 Credit Proposal." *Tax Notes* 59, no. 5 (May 3, 1993): 597-98.

Turro, John. "White House Fine Tunes Foreign Provisions; Sticks to 936 Proposal." *Tax Notes* 59, no. 6 (May 10, 1993): 724.

Turro, John. "U.S. Releases Final Transfer Pricing Regulations Under Section 482." 9 *Tax Notes Int'l* 79 (July 11, 1994).

Turro, John. "Proposed Section 936 Profit Split Rule Contravenes Arm's Length Principle, Witnessess Say." 9 Tax Notes Int'l 129 (July 18, 1994).

U.S. Congressional Budget Office. "Potential Economic Impacts of Changes in Puerto Rico's Status under S.712." Report prepared for the U.S. Senate Committee on Finance, April 1990.

- U.S. Department of Commerce, Bureau of the Census, Statistical Abstract of the United States, 1975-1985.
- U.S. Department of Commerce. Guidebook: Caribbean Basin Initiative. 1991.
- U.S. Department of the Treasury. "The Operations and Effect of the Possessions Corporation System of Taxation, Sixth Report." March 1989.
- U.S. Department of the Treasury, Internal Revenue Service. "Intercompany Transfer Pricing and Cost Sharing Regulations under Section 482." 26 CFR Part 1, January 30, 1992.
- U.S. Department of the Treasury, Internal Revenue Service. "Intercompany Transfer Pricing and Cost Sharing Regulations under Section 482, Hearing." 26 CFR Part 1, July 30, 1992.
- U.S. General Accounting Office. "Pharmaceutical Industry: Tax Benefits of Operating in Puerto Rico." Briefing Report to the Chairman, Special Committee on Aging, U.S. Senate. May 1992.
- U.S. House of Representatives. "Report of the Committee on Ways and Means, U.S. House of Representatives, on H.R. 10612." Report No. 94-658, November 12, 1975.
- U.S. International Trade Commission. "Annual Report on the Impact of the Economic Recovery Act on U.S. Industries and Consumers, Second Report 1986." September 1987.
- U.S. Senate. "Report of the Committee on Finance, United States Senate, on H.R. 10612." Report No. 94-938, June 10, 1976.
- U.S. Senate, Committee on Finance. Tax Reform Act of 1986. S. Rep. No. 313, 99th Cong., 2d Sess. 384 (1986).

Wright, Delores R. "Living with the Proposed S. 482 Regulations." Tax Management Memorandum 33, no. 8 (April 20, 1992): S3-10. ◆



Date:	Time:	Pages:
To: TAX ANALYSTS SUBSCRIPTION DEPARTMEN	Т	
From:		
Company Name:		***************************************
Contact:		
Address:		
Telephone Number:		
Fax Number:		
Yes! I would like to read <i>Tax Notes International</i> in today. I understand I will receive 52 issues, each prodevelopments, for US \$749.	the coming year. Please enter moviding comprehensive coverage	ny one-year subscription e of international tax
Guarantee: If I am dissatisfied at any time within the first full refund of my payments. Thereafter, I am entitled to a p	-	, I may cancel and receive a
Customer Service: (800) 955-2444 or (800) 955-3444 within the	continental U.S., or (703) 533-460	00 outside the U.S.

THY. (702) 522 4625



TTY: (703) 533-4625

Delivery: Tax Notes International is delivered by first class mail, hand delivery, or international air mail without additional charge.

**Price:** The annual subscription (52 issues) is US \$749. Subscribers may pay for subscriptions and other services either in U.S. dollars or in one of the currencies specified below. There is a premium of 2.5-percent for payment in non-U.S. currencies. This premium is reflected in the prices set forth below. Billings reflect the exchange rates in effect when the invoices are first sent: they are not adjusted thereafter.

Prices in foreign currencies: Australia (A\$1,043); Austria (A\$ 8,284); Belgium (BF 24,253); Canada (C\$ 1,058); Denmark (DKK 4,658); Finland (FIM 3,846); France (FFr 4,045); Germany (DM 1,177); Hong Kong (HK\$ 5,933); Ireland (IR£ 500); Japan (¥75,276); Luxembourg (LFr 24,253); Malaysia (M\$ 1,958); Netherlands (Dfl 1,322); New Zealand (NZ\$ 1,279); Norway (NK¥ 5,176); Singapore (\$1,153); South Africa (R 2,748); Spain (Pta 98,123); Sweden (SEK 5,830); Switzerland (SFr 993); U.K. (£494); U.S. (US \$749).